

Document de treball de l'IEB 2015/1

OVERLAPPING POLITICAL BUDGET CYCLES IN THE LEGISLATIVE AND THE EXECUTIVE

Dirk Foremny, Ronny Freier, Marc-Daniel Moessinger, Mustafa Yeter

Fiscal Federalism

Document de treball de l'IEB

OVERLAPPING POLITICAL BUDGET CYCLES IN THE LEGISLATIVE AND THE EXECUTIVE

Dirk Foremny, Ronny Freier, Marc-Daniel Moessinger, Mustafa Yeter

The **IEB** research program in **Fiscal Federalism** aims at promoting research in the public finance issues that arise in decentralized countries. Special emphasis is put on applied research and on work that tries to shed light on policy-design issues. Research that is particularly policy-relevant from a Spanish perspective is given special consideration. Disseminating research findings to a broader audience is also an aim of the program. The program enjoys the support from the **IEB-Foundation** and the **IEB-UB** Chair in Fiscal Federalism funded by Fundación ICO, Instituto de Estudios Fiscales and Institut d'Estudis Autonòmics.

The **Barcelona Institute of Economics (IEB)** is a research centre at the University of Barcelona (UB) which specializes in the field of applied economics. The **IEB** is a foundation funded by the following institutions: Applus, Abertis, Ajuntament de Barcelona, Diputació de Barcelona, Gas Natural, La Caixa and Universitat de Barcelona.

Postal Address: Institut d'Economia de Barcelona Facultat d'Economia i Empresa Universitat de Barcelona C/ Tinent Coronel Valenzuela, 1-11 (08034) Barcelona, Spain Tel.: + 34 93 403 46 46 Fax: + 34 93 403 98 32 ieb@ub.edu http://www.ieb.ub.edu

The IEB working papers represent ongoing research that is circulated to encourage discussion and has not undergone a peer review process. Any opinions expressed here are those of the author(s) and not those of IEB.

OVERLAPPING POLITICAL BUDGET CYCLES IN THE LEGISLATIVE AND THE EXECUTIVE *

Dirk Foremny, Ronny Freier, Marc-Daniel Moessinger, Mustafa Yeter

ABSTRACT: We advance the literature on political budget cycles by testing separately for cycles in expenditures for elections in the legislative and the executive. Using municipal data, we can separately identify these cycles and account for general year effects. For the executive branch, we show that it is important whether the incumbent re-runs. To account for the potential endogeneity associated with this decision, we apply a unique instrumental variables approach based on age and pension eligibility rules. We find sizable and significant effects in expenditures before council elections and before joint elections when the incumbent re-runs.

JEL Codes: H11, H71, H72, H74

Keywords: Election cycles, municipal expenditures, council and mayor elections, instrumental variables approach

Dirk Foremny Universidad de Barcelona & IEB Av. Diagonal, 690, Torre 6, planta 1 08034 Barcelona, Spain E-mail: <u>foremny@ub.edu</u>

Marc-Daniel Moessinger Centre for European Economic Research (ZEW Mannheim) 68161 Mannheim, Germany E-Mail: <u>moessinger@zew.de</u> Ronny Freier German Institute for Economic Research (DIW Berlin) and Free University Berlin Mohrenstrasse 58 G-10117 Berlin E-Mail: rfreier@diw.de

Mustafa Yeter Centre for European Economic Research (ZEW Mannheim) 68161 Mannheim, Germany E-Mail: yeter@zew.de

We would like to thank Florian Ade, Peter Haan, Niklas Potrafke, Friedrich Schneider, Johannes Voget as well as seminar participants at the ZEW Mannheim for helpful comments and suggestions. We are greatly indebted to Fabian Franke, Jenny Freier, Kathrin Friedrich, Heike Hauswald, Ilknur Köysüren, Larissa Lehn and Marc Schmidt who provided excellent research assistance. Ronny Freier (project: 10.12.2.092) as well as Marc-Daniel Moessinger and Mustafa Yeter (project: 10.12.1.094) gratefully acknowledge financial support from the Fritz Thyssen foundation. Dirk Foremny gratefully acknowledges financial support from projects ECO2012-37131 (Ministerio de Educacíon y Ciencia) and 2014SGR420 (Generalitat de Catalunya). The usual disclaimer applies. The authors declare that they do not have relevant or material financial interests that relate to the research described in this paper.

1. Introduction

Originating with Nordhaus (1975), a main paradigm of the political economics literature is that political agents act rational when deliberately manipulating the economy or particular fiscal aggregates to gain an electoral advantage. Further theory (e.g., Rogoff and Sibert, 1988 and Rogoff, 1990) has confirmed the main prediction of this paradigm even under rational expectations of the voters.¹ While the early empirical literature looked for cycles in macroeconomic aggregates such as unemployment and inflation², more recent work high-lighted credible evidence for political budget cycles (henceforth, PBC) in fiscal aggregates such as debt, expenditures, and its composition (see Keech and Pak, 1989, Alesina et al., 1997, or Brender and Drazen, 2005).³ Overall, the phenomenon of PBC is relatively well understood and the empirical evidence is highlighting the relevance in actual policy making.

This paper aims to contribute to this literature in two ways. First, we study PBCs in expenditures for elections in the legislative *and* the executive. The existing literature mostly had to focus on the legislative branch or on joint elections. Our data and institutional design provides us with the unique opportunity to study the cycles of the two institutions separately and evaluate the effects of overlaps. Second, for the analysis of PBCs in the executive, we highlight the concerns related to the endogenous decision of incumbent executives to rerun. Overcoming this endogeneity is crucial as we expect PBCs only when the incumbent competes for re-election. Here, we use an instrumental variables approach based on the age of the incumbent as well as on pension eligibility rules to achieve credible identification. To the best of our knowledge, both innovations have not been treated in the literature.

The testing ground for our analysis is the German municipal level. Our paper is therefore related to the literature that focuses on the sub-national level. In Table A.1 in the appendix, we provide a detailed and exhaustive overview of papers studying PBCs in expenditures at the sub-national level which relate directly to our first contribution. Evidence for cycles in expenditures has been documented on state, regional and municipal level in twelve different countries. Most studies focus on legislative elections and use variation over time to separate election effects from general year effects.⁴ Apart from the papers in Table A.1, we identified

¹ Alternatively, Persson and Tabellini (2000), and Shi and Svensson (2006) provide a theoretical foundation for political budget cycles under rational expectations and moral hazard. Also, a second strand of theoretical work relates to partian cycles (e.g., Hibbs, 1977; Alesina, 1987, 1988a) which explains electoral cycles by shifts in political ideology.

² Studies finding only weak or no evidence for cycles in real macroeconomic aggregates include, among others, Lächler (1978), Golden and Poterba (1980) and Alesina et al. (1997).

³ Further evidence at the national level is provided by Alesina (1988b), Alesina et al. (1992), Schuknecht (1999), Potrafke (2012), and Klomp and De Haan (2013).

⁴ Note that the literature here is divided into papers that study the size of aggregate expenditures close to elections (similar to our question) and several authors testing for PBCs in the composition of expenditures. Evidence for the latter case was first presented by Blais and Nadeau (1992) and Kneebone and Mckenzie (2001) for Canada and later by Drazen and Eslava (2010), Veiga and Veiga (2007b) and Akhmedov and Zhuravskaya (2004) in the case of municipal or regional legislative elections in Colombia, Portugal and Russia respectively. Common to all their results is a shift in expenditures especially to

three studies that are of particular importance for us. Using alternative outcome variables and focusing on council elections in Germany, Foremny and Riedel (2014) (taxes), Englmaier and Stowasser (2013) (saving banks lending) and Mechtel and Potrafke (2013) (unemployment) provide evidence that PBCs are of large significance in Germany.⁵ Furthermore, in a recent working paper, Furdas et al. (2014) analyse a related question of how PBCs evolve in different federal tiers.

For the second part of the paper, the related literature is comparatively small. Only few papers have looked at PBCs in the executive branch (e.g., Rose, 2006; Alt and Rose, 2009). The reason is that the executive is often not elected directly by the voter. And even if separate elections are held, the timing of those elections often coincides with the legislative. The two papers that are closest to our approach are by Rosenberg (1992) for Israel and Aidt et al. (2011) for Portugal. Both papers study combined municipal elections and control directly for whether the incumbent executive re-runs in the election.⁶ Our contribution here is that we treat the endogeneity of the incumbent's decision to re-run with an instrumental variables approach.

The particular data that we use come from German municipalities in the states of *Bavaria* and *Baden-Württemberg* (henceforth BAY and BW respectively) covering the period between 1992-2006. To our advantage, the two states are comparable in many dimensions. Both have had direct mayoral elections for a long time, use a comparable constitutional design governing the affairs of local municipalities and are comparable in population size as well as economic performance. For both states, we observe expenditures for the entire universe of municipalities throughout the sample period. Complementing the spending data, we collected data on elections (mayoral and council elections), other fiscal measures (state transfers) and important socio-economic characteristics (population, municipal age structure, ideological indicators for councils and mayors). For BAY, we indeed have full information on the entire set of variables. In BW, all information is available except for the mayoral elections data which we had to survey town by town (see section 3).

Crucial for the validity of our empirical analysis is the fact that state law regulates election cycles and that election dates are therefore set exogenous with very little or virtually no influence on the timing by municipalities.⁷ The council elections in BAY and in BW are held

categories with high visibility for the electorate.

⁵ Earlier studies for Germany involve Seitz (2000), Galli and Rossi (2002) as well as Schneider (2010) and test for PBCs in total expenditure and deficits. Rather than on the municipal-level, as it is the case in our analysis, these studies rely on state-level data and elections.

⁶ There is also some indirect evidence that expenditure manipulation may indeed be a good strategy for an incumbent. Freier (2011) provides evidence that the voters reward incumbent mayors that succeeded in raising the level of expenditures prior to election, but at the same time fail to punish the running mayor for increases in debt. Veiga and Veiga (2007a) confirm this pattern using Portuguese data.

⁷ In case of council elections, individual municipalities in BAY or BW have no influence whatsoever. In case of mayoral elections, exceptions are given by cases in which the town council may recall the mayor and thereby in principle can manipulate the timing of elections. This, however, only occurred in a negligible number of towns, see Kern (2008). The term of the mayor may also end prematurely due to

on state-specific dates which are identical for all municipalities in the respective state. In contrast, regulations with respect to mayoral elections differ between states. While election dates for the executive branch in BAY are aligned with council elections, BW exhibits municipal-specific dates which differ from dates of council elections (see section 4). Given this institutional structure of election dates, we can identify PBC both in mayoral as well as in council elections *and* separate those effects from general year effects.

Our main results are threefold. First, we identify sizable and significant increases in total expenditures in pre-election years for the legislative branch. These results are in the range of 1.3–1.8 percent of total expenditures and remain stable through all our robustness tests. For a middle-sized town of 10,000 inhabitants, this amounts to additional spending in the order of about 300,000 Euro in the year prior to the election. Second, we find that these fiscal cycles in council elections are unaffected by the timing of mayoral elections. Third, we show that there is an effect of overlapping cycles conditional on the incumbent's decision to re-run for office. Municipal expenditures increase in joint pre-election and election years if the incumbent mayor seeks for re-election and increase in joint post-election years if the mayor did not re-run. For instance, given that the incumbent mayor enters the race for office again, municipal spending in overlapping pre-election years is 5.5 percent higher than in the counterfactual situation when the incumbent mayor does not seek re-election.

The paper is organized as follows. Section 2 discusses the institutional setting, whereas section 3 presents our data and descriptives. In section 4 we lay out the empirical strategy, section 5 discusses the main results and section 6 presents the sensitivity analyses. Section 7 concludes.

2. Institutional background

The municipal level in Germany is the lowest of three governmental tiers.⁸ Within this structure, the municipalities are responsible for a significant share of the overall state activity.⁹ Among other things, municipalities are responsible for general administration, public order, cultural expenditures, infrastructure and public transport. Also, towns often oversee local public firms and administrate expenditures from higher-level tiers (e.g., expenditures on social welfare). Despite the often complex division of tasks between the different tiers, the municipalities (especially in BAY and BW) retain considerable discretion in spending decisions.

death or sickness (arguably exogenous) or resignation (potentially endogenous). We discuss these issues in more detail in section 4.

⁸ Besides the federal level, there exist 16 federal states and about 12,500 municipalities (organized in about 450 counties).

⁹ Municipalities in BW sign responsible for 52.9% of all expenditures in the state which is above the German average (this includes municipal budgets as well as state-own companies under the control of the local towns). Municipalities in BAY administrate 49.7% (see Federal Statistical Office, 2011).

Municipalities in Germany are governed by an elected council and an elected mayor. Councils act as the legislative body of the local town or municipality and mayors can be viewed as the executive branch.¹⁰ While councils are elected in open list proportional elections, majoritarian elections are held to determine the race for the local mayor.¹¹ In BAY, the council and the mayor are elected for six years on (generally) the same statewide election day.¹² In contrast to this, election terms of mayors and councils in BW are of different length and election days differ accordingly: the local mayor is elected for eight years, while council elections are held every five years. Furthermore, only the council elections are held on the same statewide day, whereas the election dates for mayors are municipality specific, i.e., each municipality has its own electoral cycle for mayoral elections.

We illustrate the distribution of mayoral and council elections over time in Figure 1. Within the time coverage of this study, we observe council elections held in 1996 and 2002 in BAY and in 1994, 1999, and 2004 in case of BW. In BAY, mayoral elections generally run parallel with council elections. However, some mayoral election cycles deviate from the general rule due to prematurely ended terms. This is the case for 689 mayoral elections (16% in relation to 4114 council elections). In contrast, mayoral elections in BW spread evenly throughout the years and only randomly overlap with council elections (15% in relation to 3303 council elections).¹³

Crucial for the validity of our empirical analysis is the fact that the timing of local elections (council as well as mayoral) is regulated by state law and is for the most part outside of the control of the individual municipality. The timing of the council elections is fixed entirely by given state-wide election terms and is therefore exogenous.¹⁴ While in principle this is also true for mayoral elections, the mayor's term may potentially deviate from these general rules. In particular, the mayor's term may end prematurely and necessitates new elections

¹⁰ Note that the local mayor is also a voting member of the council. Generally, the mayor is in charge of the administration, she prepares all decisions of the municipalities and oversees the implementation. Also, she is often the only full-time working politician in the town.

¹¹ The electoral rules in BAY and BW are quite similar. There are, however, also subtle differences. For council elections, both states have open-list proportional elections where voters have as many votes as there are seats in the council (which allows for vote-splitting and cross-voting). There exist no explicit hurdles for small parties. BW uses a Sainte-Lague and BAY (a two-step) D'Hondt seat allocation mechanism. For mayoral elections, both states use a two-round majoritarian procedure. If no candidate obtains the absolute majority in the first round, a second election must be held. In BAY, this second election is a classical run-up election between the two leading candidates. In BW, the second election is open (even new candidates are allowed) and first-past-the-post.

¹² There are exceptions to that rule, when mayor election terms ended prematurely. In this case, a Bavarian municipality might have an independent mayoral election for one term with the requirement to align with the state-wide election dates again in the second subsequent election.

¹³ Note that these deviations remain if we restrict the calculations to those municipal-year observations that are included in our estimations, i.e., mayoral and council election years in BW randomly overlap in 14.7%, while mayoral election years in BAY deviate from council election years in 10.7% in relation to the number of council elections.

¹⁴ There is no option for pre-mature elections for the council. As there are no formal coalition agreements, it cannot happen that a government breaks down. Also, if, a council member dies in office or resigns due to personal reasons, she is automatically replaced by a successor.

Figure 1: Number of council and mayoral elections in BAY and BW



due to a number of reasons. First, the citizens or the council may recall the mayor.¹⁵ Second, the mayor may die or resign for personal (e.g., sickness) or political reasons (lack of political support). A third reason causing deviations from the predefined length of political terms is given by existence of strict age limits inhibiting incumbents to serve beyond the age of 68 in case of BW.¹⁶

Focusing on BAY and BW in a sub-national study on PBC is attractive due to several reasons. First, historically, both states have had direct mayoral elections already since World War II. This stands in contrast to all other German states which introduced direct mayoral elections starting in the mid-1990s (see Ade and Freier, 2013). As a consequence, we cannot study individual PBCs in mayoral and council elections in other states earlier than the 2000s.¹⁷ Furthermore, since the mayor is automatically a voting member of the council in both states, studying the overlap of individual cycles is interesting, as the institutional setting may leave room for collusion between both political branches.

Second, the constitutional framework that governs the affairs of the local level follows a comparable standard in BAY and BW. Both states apply a constitutional setting that is referred to as *Süddeutsche Ratsverfassung*. Important for our study, this particular constitutional setting grants extensive rights and duties to the position of the mayor. Again, this partly stands in contrast to the institutional structure in other German states, in particular when it comes to expenditures.

Third, the two states in our analysis are comparable along many other dimensions. Both are located in direct vicinity in the South of Germany. Given the time coverage of this study, both states have had conservative state rule throughout.¹⁸ Also, both states are of

¹⁵ While this option is a constitutionally guaranteed right of the citizens and the council, we almost never observe this in the time period covered in this study.

¹⁶ We further discuss the importance of premature mayoral elections in our section on identification (see section 4).

¹⁷ And even then, evidence from those other German states would be particular as these mayoral elections are only newly introduced.

¹⁸ BAY have had absolute majorities of the conservative party throughout the time coverage of this study.

comparable size and population.¹⁹ Finally, BAY and BW are of comparable wealth and industrial performance.²⁰ In both states, the municipalities are financially well off, the level of debt is comparatively low and the towns enjoy extensive financial liberties.²¹

3. Data

We use data from municipalities in BAY and BW for the period from 1992 to 2006.²² We observe 2,056 municipalities and a population of 12.5 million inhabitants in BAY and 10.5 million inhabitants live in 1,101 municipalities in BW.

We obtained information on our outcome variable total expenditures, state transfers, results for the council elections, information on local population as well as population structure from the respective state statistical offices. We calculate municipal expenditures net of transfers, i.e., we subtract the amount of transfers from the respective state government from gross expenditures. This is crucial in order to capture and isolate the discretionary amount of spending by municipalities and to get rid of confounding influences such as spending which is initiated by an upper level governmental tier and is only administered at the municipal level. Beyond that, this transformation helps us to abstract from PBCs in the state level elections, which potentially could otherwise alter municipal budgets through transfers (see Furdas et al., 2014).

From the data for the council elections, we create dummy variables that identify a council pre-election, election or post-election year to capture the legislative PBC. Furthermore, we classify councils by the share of left parties (party members of the social democrats (SPD) and the green party (Grüne)). We summarise all relevant descriptive information in Table 1. We further explain our variable coding in Table A.2 in the appendix.

In BW, the conservative party also headed the state government, however, they did so as the predominant party in a coalition.

¹⁹ As a consequence, also the municipal structure is quite comparable. While BW has slightly larger municipalities on average, these differences are rather small compared to structures in other German states. Together, the two states account for more than 25% of the German population.

²⁰ Together they account for approximately 75% of the total state redistribution in the German fiscal equalization scheme (see Heinemann et al., 2013).

²¹ Municipalities in other German states might have very limited financial opportunities for extraordinary investment projects or other general pork barrel spending before elections.

²² The time coverage of this study is limited by the availability of data on municipal spending (*Jahresrechnungsstatistik*). Also, data for earlier mayoral elections in BW is hard to obtain.

Variable	Observations	Mean	Std. Dev.	25 %ile	Median	75 %ile	Minimum	Maximum
Total expenditures in logs (net of transfers) Council elections	39554	15.509	1.145	14.686	15.364	16.186	11.689	22.557
Pre-election year	39554	0.148	0.355	0	0	0	0	1
Election year	39554	0.153	0.360	0	0	0	0	1
Post-election year	39554	0.155	0.361	0	0	0	0	1
Mayor elections								
Pre-election year	39554	0.130	0.336	0	0	0	0	1
Election year	39554	0.134	0.340	0	0	0	0	1
Post-election year	39554	0.132	0.339	0	0	0	0	1
Dummy incumbent runs again	39554	0.739	0.439	0	1	1	0	1
Dummy full time mayor	39554	0.608	0.488	0	1	1	0	1
Dummy left mayor	39554	0.134	0.341	0	0	0	0	1
Age of the mayor	39554	51.534	8.398	46	52	57	25	85
Share left parties in council	39554	0.138	0.153	0.000	0.083	0.250	0	0.714
Population	39554	7.124	28.438	1.732	3.141	6.135	0.096	1280.610
Population squared	39554	0.859	26.815	0.002	0.009	0.037	0.000	1639.962
Share population < 15 years	39554	0.180	0.024	0.164	0.180	0.196	0.066	0.343
Share population ≥ 65 years	39554	0.171	0.046	0.136	0.161	0.200	0.054	0.462
State elections								
Pre-election year	39554	0.202	0.401	0	0	0	0	1
Election year	39554	0.216	0.412	0	0	0	0	1
Post-election year	39554	0.196	0.397	0	0	0	0	1
Instruments for incumbent runs again								
Dummy mayor is older than 60 years	39363	0.312	0.463	0	0	1	0	1
Dummy mayor is pensionable	39363	0.353	0.477	0	0	1	0	1
Notes: The table shows the descriptive statist figures, median as well as the 25% and the 75 include data from 2049 municipalities in BAY	ics for the sampl % percentiles of and 968 municij	e. We pre the distri palities in	esent sample butions for BW.	the variab	andard de les indicat	viations, r ed in the	nimimum aı left column.	nd maximum In total, we

Table 1: Sample description

8

In Table A.3 in the appendix, we highlight municipal characteristics of the final sample for BW for which we ultimately obtained adequate information on mayoral elections and compare these characteristics with information on the counterfactual full universe. Compared to the total number of 1,101 municipalities and 16,515 municipal-year observations, the estimation sample comprises only 968 municipalities and significantly less years (10,531 municipal-year observations). Especially small municipalities are lost due to missing information on mayoral elections.²³ This is plausible given our research strategy that was based on a questionnaire which smaller municipalities with less municipal staff might not be able to answer. However, we argue that our results do not suffer from a bias given the small differences in the variable means between the two groups.

Obtaining comprehensive data on mayoral elections in BW is a challenging task. While complete administrative data on elections is provided for BAY by the state statistical office, such data do not exist for BW.²⁴ Due to this lack of official data, we collected data manually by contacting and surveying all 1,101 municipalities with respect to the dates of past mayoral elections. Using obtained responses, we then searched regional newspapers as well as official announcements (*Staatsanzeiger Baden-Württemberg*) to identify further information on the specific election, e.g., the name, age, party affiliation and regular occupation of the elected mayor as well as information on whether the mayor has ran for re-election. Finally, we conducted an intensive internet search on official websites and free internet resources to fill further gaps in our data set.

Similar to council elections, we code a set of dummy variables for mayoral elections indicating the pre-election, election and post-election year in each municipality. Furthermore, we construct dummy variables for mayors indicating a membership of a left-wing party (member of the social democrats (SPD) or the green party (Grüne)), whether the mayor has a full-time position and whether the incumbent mayor re-runs for office in the current election. As the incentives to strategically manipulate public spending should especially be higher in case of the latter, we will construct respective interactions in order to capture the impact on the formation of PBCs. Finally, we include the mayor's age as a further control variable in our regressions which will also be relevant against the background of our instrumental variables approach.

As mentioned above, mayoral cycles may end prematurely due to resignation or death of the mayor. As this may potentially pose problems for the identification of electoral cycles, we exclude non-completed or prematurely ended political terms from our baseline specification. When testing the robustness of our main results, we also run estimations in which we include

 $^{^{23}}$ The p-values of the mean comparison *t*-test further show that the difference in the means is statistically significant, i.e., we cannot reject the null hypothesis that there is no difference in the means of the control variables.

²⁴ We contacted all major authorities such as the state statistical office, the association of municipalities, the state ministry of interior affairs and the state election office. No summary of official data exists for mayoral elections in BW.

all available observations including non-completed terms.

4. Identification

We use the following baseline model to identify political budget cycles in expenditures:

$$log(Y_{i,t}) = \mathbf{t}_{i,\mathbf{t}}^{\prime \mathbf{c}} \boldsymbol{\delta} + \mathbf{t}_{i,\mathbf{t}}^{\prime \mathbf{m}} \boldsymbol{\phi} + I_{i,t} \boldsymbol{\theta} + \sum_{k=-1}^{1} (t_{i,t+k}^{c} \times t_{i,t+k}^{m}) \beta_{k+2} + \sum_{k=-1}^{1} (t_{i,t+k}^{m} \times I_{i,t}) \tau_{k+2} + \sum_{k=-1}^{1} (t_{i,t+k}^{c} \times t_{i,t+k}^{m} \times I_{i,t}) \sigma_{k+2} + \mathbf{X}_{i,\mathbf{t}}^{\prime} \boldsymbol{\gamma}$$

$$+ \lambda_{t} + \mu_{i} + h(t,s) + \epsilon_{i,t}.$$

$$(1)$$

The dependent variable is the natural logarithm of municipal total expenditures, net of transfers, $Y_{i,t}$. The vectors $\mathbf{t'^c}$ and $\mathbf{t'^m}$ contain dummies capturing election dates in council (c) and mayoral (m) elections, respectively. In particular, we define these dummies for respective elections j = c, m as

$$\mathbf{t'^{j}} = \begin{bmatrix} t_{t-1}^{j} \\ t_{t}^{j} \\ t_{t+1}^{j} \end{bmatrix} \text{ and } \begin{cases} =1 \text{ in the pre-election year, 0 otherwise} \\ =1 \text{ in the election year, 0 otherwise} \\ =1 \text{ in the post-election year, 0 otherwise.} \end{cases}$$
(2)

To test for differences in spending if cycles overlap, we estimate interaction terms for congruent pre-, post-, and election years as in Foremny and Riedel (2014). The dummy variable $I_{i,t}$ denotes whether the incumbent mayor re-runs for office at the next election. The indicator is set to zero/one for the whole election period, i.e., we assume that the mayor already knows whether she will re-run for office at the end of her term when she takes office.²⁵ As a result, the estimation coefficient of the incumbent indicator only reveals spending differences between electoral periods. The estimation of interaction terms with the mayoral election dummies then allows separating general from specific incumbent effects in times of elections. Finally, the triple interactions capture effects of overlapping cycles conditional on the incumbent's decision to re-run for office.

We include general year (λ_t) and municipality (μ_i) fixed effects into the model and allow for state-specific linear and quadratic time trends which we denote by h(t, s). Additional control variables are denoted by $\mathbf{X}'_{i,t}$ and include population and population squared to control for non-linear effects of local population size, population structure indicators, state

²⁵ We assume that the mayor, at least, has some preferences about this decision which lead to the final decision at the end of her term. Note that, without this assumption, we would condition the incumbent dummy on election years which would render separating general from specific incumbent effects unfeasible.

election indicators, ideological proxies for both the mayor and the council, the age of the mayor and a dummy variable indicating whether the mayor is a full-time politician.

It is important to highlight again how our estimation specification identifies PBCs in council and mayoral elections. As council elections in BAY and BW are held in different years, the effects of council cycles can be separated from general year effects and from state-specific time trends. In contrast, the identification for the mayoral elections is more involved. As highlighted in section 2, mayoral elections in BAY are generally held at the same time as council elections with few exceptions. In BW, however, we observe mayoral elections across all years which are independent from council elections. This is the source of variation, including overlapping as well as non-overlapping elections, we exploit in order to identify individual mayoral election cycles.

A further important condition for the identification of cycles is the exogeneity of election dates. In case of council elections, terms are entirely predetermined by state regulation and cannot be influenced by individual municipalities. While for mayoral elections this is generally true as well, the municipality (and/or the mayor) may influence the timing of elections within limits. In particular, the term of the mayor may end prematurely when 1) the citizens or the council recall a mayor (which is an extremely rare event), 2) the mayor dies or resigns due to sickness (arguably exogenous to our application) or 3) the mayor resigns for other reasons (personal or political reasons).²⁶ In our baseline specification, we therefore decided to rely exclusively on fully completed mayoral election terms. This ensures that election dates are exogenous and predetermined by state regulation only (eight years in BW and six years in BAY). In the robustness section, we test the sensitivity of our results by including terms that ended prematurely. The results remain similar both in terms of size and significance.

Another issue that might be raised with respect to endogeneity is given by the incumbent's decision to re-run for office. In particular, if this decision is directly linked to the fiscal conditions of the municipality and the availability of sufficient financial means for strategic spending, estimates could not be interpreted as causal. For instance, if a fiscal shock hits the municipality shortly before an election absorbing available financial means initially intended for strategic spending and ultimately inducing the incumbent not to seek re-election, we would falsely attribute pre-election variation in spending to mayoral cycles in cases of incumbents not re-running for office.²⁷ To deal with this potential endogeneity formally, we follow an instrumental variables approach.

²⁶ It is clear that a mayor will not return to office if the term ends prematurely due to early resignation. One main reason for resignations in our data is given by mayors taking office in a larger city involving a higher remuneration. While the decision to run for office in another municipality is potentially endogenous, the timing is again exogenous as it is predetermined by the timing of elections in other towns.

²⁷ Note that a potential incompetency of the mayor would not cause such a pattern. This is the case as incompetency might influence the municipal fiscal position negatively throughout the whole term and is not necessarily attributed to the years around an election.

To this end, we use two different instruments (explained below) to account for the potential endogeneity of the incumbent indicator. However, implementing an instrumental variables approach in the context of our estimation specification (see equation (1)) is associated with increased complexity because the incumbent variable is used both as individual as well as interaction variable. This results in the fact that separate first-stage regressions are necessary to instrument for the single variable and the interaction terms. The procedure is exemplary shown in Equations (3.1) to (4) for the interaction of the incumbent dummy and the mayoral election dummies. In a first step, we separately regress the incumbent dummy (equation (3.1)) and its interactions with the mayoral election dummies (equation (3.2)) on the set of instruments, i.e., the single instruments $\mathbf{Z}_{i,t}$, their interactions with the mayoral election dummies ($\mathbf{t}_{i,t}^{m} \times \mathbf{Z}_{i,t}$), and the full set of the remaining covariates $\mathbf{C}'_{i,t}$. In a second step, we use the predicted values for the incumbent indicator and its interactions as regressors in the second stage regression (equation (4), see Angrist and Pischke (2009, p. 190ff) and Wooldridge (2010, p. 267f) for more information).²⁸

$$I_{i,t} = \mathbf{t}_{i,t}^{\prime \mathbf{c}} \boldsymbol{\alpha}_1 + \mathbf{t}_{i,t}^{\prime \mathbf{m}} \boldsymbol{\alpha}_2 + \mathbf{Z}_{i,t} \boldsymbol{\alpha}_3 + (\mathbf{t}_{i,t}^{\prime \mathbf{m}} \times \mathbf{Z}_{i,t}) \boldsymbol{\alpha}_4 + \mathbf{C}_{i,t}^{\prime} \boldsymbol{\alpha}_5 + \xi_{i,t}$$
(3.1)

$$(\mathbf{t}_{\mathbf{i},\mathbf{t}}^{\prime \mathbf{m}} \times \mathbf{I}_{\mathbf{i},\mathbf{t}}) \equiv \mathbf{T}\mathbf{I}_{\mathbf{i},\mathbf{t}} = \mathbf{t}_{\mathbf{i},\mathbf{t}}^{\prime \mathbf{c}} \boldsymbol{\kappa}_{1} + \mathbf{t}_{\mathbf{i},\mathbf{t}}^{\prime \mathbf{m}} \boldsymbol{\kappa}_{2} + \mathbf{Z}_{\mathbf{i},\mathbf{t}} \boldsymbol{\kappa}_{3} + (\mathbf{t}_{\mathbf{i},\mathbf{t}}^{\prime \mathbf{m}} \times \mathbf{Z}_{\mathbf{i},\mathbf{t}}) \boldsymbol{\kappa}_{4} + \mathbf{C}_{\mathbf{i},\mathbf{t}}^{\prime} \boldsymbol{\kappa}_{5} + \boldsymbol{\varsigma}_{i,t}$$
(3.2)

$$log(Y_{i,t}) = \mathbf{t}_{i,\mathbf{t}}^{\prime \mathbf{c}} \boldsymbol{\delta} + \mathbf{t}_{i,\mathbf{t}}^{\prime \mathbf{m}} \boldsymbol{\phi} + \widehat{I}_{i,t} \boldsymbol{\theta} + \widehat{\mathbf{TI}}_{i,\mathbf{t}} \boldsymbol{\pi} + \mathbf{X}_{i,\mathbf{t}}^{\prime} \boldsymbol{\gamma} + \lambda_t + \mu_i + h(t,s) + \epsilon_{i,t}$$
(4)

Fortunately, the institutional design allows us to construct a set of instruments to get consistent estimates, as they are uncorrelated with the error term of equation (1), but highly correlated with the mayor's decision to re-run for office. We use the following variables and their interactions with the election dummies to instrument the incumbent indicator, all of which we derived from a careful exploitation of the features of the municipal electoral law:²⁹

First, we create a dummy variable which is set to one if the mayor is eligible to receive pensions (*dummy mayor is pensionable*). In both states, in general, the mayor is only eligible to receive pensions if she has served as a temporary civil servant for a particular number of years and has completed at least one entire electoral term.³⁰

²⁸ One might falsely consider to estimate the first stage regression exclusively for the single instrument and the incumbent dummy and subsequently compute the interaction term for the second stage regression manually by using the predicted values of the incumbent indicator. This procedure, however, will produce inconsistent estimates. The results of the first stage regression for the incumbent dummy is presented in Table A.5 in the appendix. The results of the remaining first stage regressions for the interaction terms are available upon request. $\mathbf{C}'_{i,t} \equiv \mathbf{X}'_{i,t} + \lambda_t + \mu_i + h(t,s)$.

²⁹ Again, as for the incumbent runs again dummy variable, both indicators are equal to zero/one for the whole election period.

³⁰ In BAY, the mayor is eligible to receive pensions if she has served for two complete election periods and was not re-elected at the next election or refused to accept the election. In BW, the ruling is more complex: The mayor is eligible to receive pensions if she has completed the entire election period and a) has served for 18 years and is older than 47 years in the year of the election, b) has served for two electoral terms or c) has served for eight years and is older than 60 years in the year of the election. For

Second, we create a dummy variable which is equal to one if the mayor is older than 60 years in the year of the mayoral election (*dummy mayor is older than 60 years*). This threshold is motivated by the age distribution of incumbent mayors not re-running for office in the year of election (see Figure 2). There is a significant break in the distribution at the age of 60 years, i.e., mayors older than 60 years are less likely to seek re-election.



Figure 2: Age distribution for mayors who do not rerun for office in the election year

The two instruments we constructed fulfill both necessary conditions for an instrumental variables regression: they are relevant in terms of explaining the mayor's decision to run for re-election (mayors above the age of 60 years and mayors who are eligible to receive pensions have less incentives to run for re-election) but are exogenous to local expenditures. In case of the latter, there is no direct link between pension eligibility and public spending. With respect to the mayor's age, in contrast, one might argue that older mayors might have different preferences shaping their spending behavior. However, we account for this possibility by directly controlling for the mayor's age in the regressions. Note that this indicator is insignificant throughout all specifications (see Table A.4 in the appendix).

5. Results

Tables 2 to 4 present the main results of our estimations. Results of the fixed effects OLS regression are reported in Table 2, while Table 3 presents the results for the instrumental variables approach. In both tables, we only show the results for our main variables of interest (council and mayoral election dummies, the incumbent runs again dummy as well as the various interaction effects). The results for the control variables are presented in the

further details see Article 21 KWBG (*Law on local elections and public servants*) (BAY) and §36ff LBG (*Law on state public servants*) (BW).

appendix (see Table A.4).³¹ As some of our hypotheses are based on interaction terms and the respective sum of individual coefficients, we refer to Table 4 which presents F-tests of joint significance for the (combination of) estimated effects and is therefore complementary to the other two tables.

Tables 2 and 3 are structured in the same way: in columns (1) and (2), we specify simple PBC models in which we study the council and mayoral election effects separately. While column (1) shows the results for the council elections, effects of the mayoral elections are modeled in column (2). In column (3), we combine both cycles and estimate a joint model. In column (4), we further estimate the interactions between joint council and mayoral election years and explore effects of overlapping cycles. In column (5), we estimate the interaction of the incumbent dummy with the mayoral election dummies. Finally, we present the estimation results of the full model in column (6). Here, we estimate the effects of overlapping PBCs conditional on the incumbent's decision to re-run for office.

Council and mayor cycles

We find sizable and statistically significant election effects throughout all specifications only for the pre-election year in case of council elections. This holds true using the fixed effect OLS or the instrumental variables approach (see Tables 2 and 3).

The dependent variable is given as logarithm of municipal total expenditures (net of transfers), the coefficients must thus be interpreted as semi-elasticities. The estimated effect ranges between 1.3–1.8 percent of total expenditures (except for the result of column (6) in Table 3). For a town of 10.000 inhabitants this amounts to about 300.000 Euro in additional expenditures in the year prior to a council election. This pre-election effect for the council is robust to the inclusion of the mayoral cycle as well the various interaction effects.

For mayoral elections, we do not find evidence for a robust individual cycle in total expenditures. While we do find evidence for unconditional post-election year cycles in mayoral elections (column (2) and (3) in Tables 2 and 3), this effect disappears when further interaction effects are included. Remaining point estimates for the election dummies in case of mayoral elections are quite small and statistically insignificant.

Overlapping cycles and incumbent effects

We find heterogeneous evidence in case of overlapping cycles (column 4). While neither point estimates nor F-tests on joint significance point to cycles in cases of pre-election and election years, post-election years are the exception. Here, municipal expenditures seem to increase

³¹ Interestingly, neither the ideology proxies nor the full-time mayor dummy and the mayor's age enter the model with statistically significant effects. While increasing population has a concave effect on local expenditures, the shares of young and old people do not affect spending. The estimation results for the control variables of the instrumental variables regression are available upon request.

in the year after simultaneous elections in both political branches. The mayor post-election coefficients are 0.014^{**} (p-value 0.041, fixed effects OLS regression) and 0.015^{**} (p-value 0.035, instrumental variables regression). This points to a potential collusion between both political branches in order to gain an electoral advantage.

However, the underlying channel though remains unclear in these estimations. We therefore extend the model and include information on whether the incumbent mayor seeks reelection in this municipality in order to explore one potential mechanism behind this result. Therefore, in columns (5) and (6) of either table we augment the model by interactions of a dummy including information whether or not the incumbent mayor seeks for re-election with the cycle variables.

The fact whether the incumbent re-runs for office exhibits a statistically significant and positive unconditional impact on municipal expenditures in case of the fixed effects OLS regressions (see Table 2). However, this result is not confirmed by the instrumental variables approach (Table 3). While point estimates increase in case of the instrumental variables regression, coefficients become marginally insignificant. Nonetheless, the results indicate unconditional higher spending in election periods in which the incumbent is seeking re-election.

To asses the quantitative and qualitative results of columns (5) and (6) in case of multiple interaction terms, we refer to Table 4 which contrasts the results for both columns in case of the fixed effects OLS and the instrumental variables regression. Note that Table 4 reports p-values of underlying F-tests in parantheses. The structure of the table is as follows: while distinguishing between pre-election, election, and post-election years, we compute marginal effects for overlapping council and mayoral elections conditional on whether the incumbent mayor runs for re-election. We then test whether the difference between these effects is statistically different from zero. Furthermore, we compute the marginal incumbent effect conditional on whether there is a simultaneous election. This effect can then be contrasted to the unconditional effects shown in the two baseline tables.

We find a sizable and statistically significant difference in pre-election years for mayors conditional on whether the incumbent re-runs for office (line *Difference D-C*). If the mayor does not run for re-election, local expenditures decrease in the year before the mayoral election. In contrast, if the mayor re-runs, the change in expenditures turns positive with a maximum difference of 5.5 percent (column 4). Furthermore, we find a sizable and significant incumbent effect in the range of 3.2 - 3.8 percent (line *E*) if the mayor re-runs conditional on joint council and mayoral elections in the next year. Note that this effect is much higher than the unconditional point estimates presented in Tables 2 and 3.

As indicated before, results are heterogeneous with respect to the timing of elections. In case of election years, we only find a statistically significant difference for the council but not for mayors. More precisely, if a council election takes place and there is a mayoral

	Depend (1)	lent variable: (2)	total expen (3)	ditures in log (4)	gs (net of trat (5)	nsfers) (6)
Council elections						
Pre-election year	0.015^{**} [2.398]		0.013^{**} [2.090]	0.014^{**} [2.141]	0.014^{**} [2.246]	0.018^{*} [1.691]
Election year	$0.002 \\ [0.257]$		-0.002 [-0.296]	0.000 [0.037]	-0.002 [-0.251]	-0.003 [-0.268]
Post-election year	0.005 [0.996]		$0.000 \\ [0.011]$	-0.004 $[-0.547]$	0.000 [0.088]	-0.007 [-0.672]
Mayor elections Pre-election year		0.003 [0.876]	$0.001 \\ [0.193]$	$0.002 \\ [0.462]$	-0.018*** [-2.590]	-0.003 [-0.310]
Election year		$0.006 \\ [1.442]$	$0.006 \\ [1.511]$	$0.007 \\ [1.310]$	-0.007 $[-0.865]$	$0.009 \\ [0.804]$
Post-election year		0.009^{***} [2.612]	0.009^{**} [2.420]	$0.006 \\ [1.251]$	-0.005 [-0.626]	-0.009 [-0.775]
Mayor elections \times council elections Joint pre-election year				-0.005 $[-0.518]$		-0.023 [-1.344]
Joint election year				-0.004 [-0.452]		-0.022 [-1.370]
Joint post-election year				$0.008 \\ [0.865]$		0.012 [0.775]
Mayor elections \times incumbent runs again Pre-election year					0.024^{***} [3.209]	0.007 $[0.627]$
Election year					0.017^{*} [1.763]	-0.005 $[-0.344]$
Post-election year					0.019^{*} [1.940]	0.020 [1.505]
Council elections \times incumbent runs again Pre-election year						-0.003 [-0.263]
Election year						$0.005 \\ [0.493]$
Post-election year						$0.005 \\ [0.508]$
Joint elections \times incumbent runs again Pre-election years						0.024 [1.301]
Election years						$0.025 \\ [1.410]$
Post-election years						-0.006 [-0.388]
Incumbent runs again dummy	0.013^{***} [2.940]	0.013^{***} [3.053]	0.013^{***} [3.050]	0.013^{***} [3.043]	$0.004 \\ [0.718]$	$0.004 \\ [0.631]$
Control variables	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
Observations R^2 Cluster	$39554 \\ 0.077 \\ 3017$	$39554 \\ 0.077 \\ 3017$	$39554 \\ 0.077 \\ 3017$	$39554 \\ 0.077 \\ 3017$	$39554 \\ 0.077 \\ 3017$	$39554 \\ 0.077 \\ 3017$

Table 2: PBCs in public expenditure (fixed effects OLS regression)

Notes: All specifications include municipality and time fixed effects. Robust t-values are shown in brackets. ***, (**), (*) denotes significance at the 1-, (5-), (10-)%-level. Error terms are clustered on the municipal level. Interaction effects for election dummies are estimated only for congruent election years. F-Tests for the interaction effects in columns (5) and (6) are shown in Table 4. The estimation also includes a number of additional control variables, see full list in Table A.4 in the appendix. Source: Own calculations.

	Depend (1)	ent variable: (2)	total exper (3)	ditures in 1 (4)	logs (net of (5)	transfers) (6)
Council elections Pre-election year	0.016^{**} [2.477]		0.014^{**} [2.194]	0.015^{**} [2.266]	0.015^{**} [2.293]	0.044^{**} [2.103]
Election year	0.002 [0.282]		-0.001 [-0.140]	0.001 [0.138]	-0.001 [-0.125]	0.012 [0.639]
Post-election year	0.005 [0.918]		-0.001 [-0.145]	-0.004 $[-0.655]$	-0.000 [-0.002]	-0.003 [-0.216]
Mayor elections Pre-election year		0.003 [0.786]	0.000 $[0.066]$	$0.002 \\ [0.401]$	-0.017 [-1.459]	0.000 [0.017]
Election year		$0.004 \\ [1.126]$	$0.005 \\ [1.129]$	$0.006 \\ [0.978]$	0.003 [0.181]	0.036^{*} [1.726]
Post-election year		0.010^{***} [2.641]	0.010^{**} [2.475]	0.007 [1.310]	0.033 [1.330]	-0.003 [-0.138]
Mayor elections \times council elections Joint pre-election year				-0.005 $[-0.568]$		-0.049* [-1.650]
Joint election year				-0.004 [-0.375]		-0.046* [-1.854]
Joint post-election year				0.008 [0.855]		0.068^{**} [2.480]
Mayor elections \times incumbent runs again Pre-election year					$0.020 \\ [1.511]$	0.001 [0.065]
Election year					0.002 [0.084]	-0.041 $[-1.599]$
Post-election year					-0.033 $[-1.036]$	0.013 [0.491]
Council elections \times incumbent runs again Pre-election year						-0.034 [-1.413]
Election year						-0.013 [-0.611]
Post-election year						-0.001 [-0.042]
Joint elections \times incumbent runs again Pre-election years						$0.054 \\ [1.547]$
Election years						0.057^{**} [1.985]
Post-election years						-0.080 ^{**} [-2.293]
Incumbent runs again dummy	0.015 [1.489]	0.017 [1.635]	$0.017 \\ [1.617]$	0.017 [1.613]	0.012 [0.791]	0.017 $[1.111]$
Control variables	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
Observations R^2 Cluster Underidentification test (Kleibergen-Paap rk LM) Weak identification test (Kleibergen-Paap rk Wald F) Overidentification test (Hansen J p-value)	$\begin{array}{r} 39344 \\ 0.077 \\ 2997 \\ 888.7 \\ 1066.3 \\ 0.824 \end{array}$	$\begin{array}{c} 39344 \\ 0.077 \\ 2997 \\ 865.4 \\ 1024.9 \\ 0.851 \end{array}$	$39344 \\ 0.077 \\ 2997 \\ 864.3 \\ 1023.0 \\ 0.850$	$\begin{array}{c} 39344 \\ 0.077 \\ 2997 \\ 863.4 \\ 1021.0 \\ 0.848 \end{array}$	$39344 \\ 0.076 \\ 2997 \\ 833.2 \\ 150.4 \\ 0.977$	$39344 \\ 0.075 \\ 2997 \\ 674.8 \\ 49.2 \\ 0.755$

Table 3: PBCs in public expenditure (instrumental variables regression)

Notes: The instrumented variable is the incumbent runs again dummy. Instruments are a dummy indicating whether the mayor is pensionable, a dummy indicating whether the mayor is older than 60 years, and the interactions of the instruments with the election indicators (if used in the regression, see the results of the first stage regression in Table A.5 in the appendix). All specifications include municipality and time fixed effects. Robust t-values are shown in brackets. *******, (******), (*****) denotes significance at the 1-, (5-), (10-)%-level. Error terms are clustered on the municipal level. Interaction effects for election dummies are estimated only for congruent election years. F-Tests for the interaction effects are available upon request). Source: Own calculations.

	Fixed eff	ects OLS	Instrume	ntal variables
	(1)	(2)	(3)	(4)
$\begin{array}{l} Pre\text{-}election \ year \\ \text{A: Council} \ \ \text{joint pre-election years, incumbent runs again} = 0 \end{array}$		-0.005 (0.695)		-0.005 (0.814)
B: Council joint pre-election years, incumbent runs again = 1		$0.016 \\ (0.115)$		$0.015 \\ (0.167)$
Difference if incumbent reruns (B-A)		$\begin{array}{c} 0.021 \\ (0.138) \end{array}$		$\begin{array}{c} 0.020 \\ (0.424) \end{array}$
C: Mayor joint pre-election years, incumbent runs again = 0	-0.018^{***} (0.010)	-0.026^{**} (0.034)	-0.017 (0.145)	-0.049^{**} (0.033)
D: Mayor joint pre-election years, incumbent runs again = 1	$0.006 \\ (0.155)$	$0.005 \\ (0.559)$	$0.003 \\ (0.431)$	$0.006 \\ (0.475)$
Difference if incumbent reruns (D-C)	0.024^{***} (0.001)	0.031^{**} (0.015)	$\begin{array}{c} 0.020 \\ (0.131) \end{array}$	0.055^{**} (0.039)
E: Incumbent runs again joint pre-election years	0.028^{***} (0.001)	0.032^{***} (0.001)	0.032^{*} (0.067)	0.038^{**} (0.045)
Election year F: Council joint election years, incumbent runs again = 0		-0.025^{*} (0.057)		-0.034^{*} (0.063)
G: Council joint election years, incumbent runs again = 1		$0.005 \\ (0.631)$		$0.010 \\ (0.353)$
Difference if incumbent reruns (G-F)		0.030^{**} (0.039)		0.044^{**} (0.047)
H: Mayor joint election years, incumbent runs again = 0	-0.007 (0.387)	-0.013 (0.327)	$0.003 \\ (0.856)$	-0.010 (0.683)
I: Mayor joint election years, incumbent runs again = 1	$\begin{array}{c} 0.010 \\ (0.032) \end{array}$	$\begin{array}{c} 0.007 \\ (0.334) \end{array}$	$\begin{array}{c} 0.005 \ (0.378) \end{array}$	$\begin{array}{c} 0.006 \\ (0.463) \end{array}$
Difference if incumbent reruns (H-I)	0.017^{*} (0.078)	$\begin{array}{c} 0.020 \\ (0.150) \end{array}$	$\begin{array}{c} 0.002 \\ (0.933) \end{array}$	$\begin{array}{c} 0.016 \\ (0.564) \end{array}$
J: Incumbent runs again joint election years	0.021^{***} (0.005)	0.029^{***} (0.002)	$\begin{array}{c} 0.014 \\ (0.296) \end{array}$	$0.020 \\ (0.175)$
Post-election year K: Council joint post-election years, incumbent runs again = 0		$0.005 \\ (0.698)$		0.065^{**} (0.018)
L: Council joint post-election years, incumbent runs again = 1		$0.004 \\ (0.684)$		-0.016 (0.202)
Difference if incumbent reruns (L-K)		-0.001 (0.906)		-0.081^{**} (0.025)
M: Mayor joint post-election years, incumbent runs again = 0	-0.005 (0.531)	$0.003 \\ (0.803)$	$0.033 \\ (0.183)$	0.065^{*} (0.051)
N: Mayor joint post-election years, incumbent runs again = 1 $$	$\begin{array}{c} 0.014^{***} \\ (0.002) \end{array}$	$0.017 \\ (0.023)$	$0.000 \\ (0.991)$	-0.002 (0.861)
Difference if incumbent reruns (N-M)	0.019^{*} (0.052)	$0.014 \\ (0.296)$	-0.033 (0.300)	-0.067 (0.111)
O: Incumbent runs again joint post-election years	0.023^{***} (0.002)	0.023^{**} (0.015)	-0.021 (0.314)	-0.051 (0.119)

Table 4: Marginal effects of interaction terms: Fixed effects OLS regression vs.instrumental variables regression

Notes: The marginal effects are computed by using the coefficients of columns (5) and (6) of Table 2 and Table 3, respectively. Two-sided p-values are shown in parantheses. ***, (**), (*) denotes significance at the 1-, (5-), (10-)%-level. Error terms are clustered on the municipal level. Source: Own calculations. election in the same year, expenditures are increased by 4.4 percent if the incumbent seeks re-election (line *Difference G-F*, column 4). While the fixed effects OLS regressions again indicate a significant impact of incumbency, the instrumental variables regressions do not confirm these results (line J).

The same pattern for the incumbent effect can be observed in post-election years. Here, the results from the fixed effects OLS regression are not confirmed by the instrumental variables approach (line O). A result that is obtained from the instrumental variables approach though is given by the difference for council elections conditional on the candidacy of incumbent mayors (line *Difference L-K*). If the mayor did not re-run in the simultaneous elections which took place in the previous year, expenditures are higher than if the mayor re-ran.

Taken together, municipal expenditures increase in joint pre-election and election years if the mayor runs for re-election and increase in joint post-election years if she did not. In case of the first two results, one might argue that these are expected as incentives for strategic spending and collusion with the council are highest if the incumbent seeks re-election. Furthermore, we observe the pattern that in these settings mayors not re-running for office usually decrease spending (e.g., line C). The third main result of a negative difference conditional on the candidacy of the incumbent mayor may be justified with shifts in budget over time rather than deficit spending. While incentives for strategic spending previous and contemporary to elections is highest, this may induce shifts of future expenditure to the present causing a decline in post-election expenditure. The observation of a positive and statistically significant council effect in post e-election years conditional on the incumbent not re-running for office (line K) may be explained by the new mayor colluding with the council to initiate investment projects promised throughout her election campaign.

6. Robustness checks

We carry out various sensitivity tests to check for the robustness of our findings (see Table 5). First, we include non-completed election periods in the estimation. These periods were excluded in the previous regressions as the timing of mayoral elections could be endogenous. Second, we exclude county-free cities from the estimation sample.³² PBCs in these cities might differ from the full sample because elections are rather partian than in smaller municipalities. Third, to check whether the results are affected by the clustering of the error terms, we alternatively cluster at the county level. Fourth, we re-estimate all specifications for data on gross expenditures, i.e., we add state transfers to municipal expenditures. This may provide further insights into the formation of PBCs as municipal spending may be co-financed by upper-level governmental tiers. Finally, we estimate our model using a single instrument rather than two and rely on the dummy indicating whether the mayor is

 $[\]overline{}^{32}$ There are 9 county-free cities in BW and 25 county-free cities in BAY.

older than 60 years in the year of the mayoral election. This is due to the Bavarian state regulation associated with pension eligibility, i.e., there might be a confounding effect as one condition to become pensionable in BAY is that the mayor has ran for re-election (but was not elected or refused to accept her election). Accordingly, a mayor could strategically decide to re-run in order to become pensionable although it is not her intention to take office after the election. However, the instrumental variables regression with only one instrument comes at the cost that we cannot test for overidentification anymore.

The robustness checks are carried out for columns (5) and (6) of Table 3. The results are reported in Table 5. The letters in italic font at the end of the lines correspond to the letters in Table 4. Only the main results of the previous findings are presented, i.e., we show the unconditional coefficient for council pre-election years, the incumbent effect conditional on joint pre-election years, and the difference in spending if the incumbent re-runs conditional on joint pre-election, election, and post-election years. The estimates include the full set of covariates. Again, p-values are shown in parentheses.

The results reveal that the previous findings are highly robust. Both, the economic and statistical significance of the unconditional council pre-election year effect and the conditional incumbent effect are confirmed. Furthermore, there are only minor changes with respect to the difference in joint pre-election, election, and post-election years conditional on whether the incumbent runs for re-election. While the algebraic sign and the point estimates only change slightly, the statistical significance is lost in council election years when we include non-completed election periods (column (2), line G-F) and in council post-election years when we carry out the regressions with only one instrument (column (4), line L-K).

Nonetheless, the main findings of our study still hold. There is an inversion in the effect for pre-election and election years compared to post-election years conditional on the incumbent's decision to re-run. Expenditures are higher in the year before the election and in the election year if the incumbent runs for re-election and cycles overlap, and are higher in post-election years if the incumbent did not run for re-election and elections did overlap.

	With non-	omnlatad tanna	Without con	intri-frag citias	Clustering	n county level	Cross ave	andituree	Only one	net rum ent
	(1)	(2)	(3)	(4)	(5)	(6)	(2)	(8)	(9)	(10)
Conneil nue-election view	×* LO O	0.037*	0.010 * *	10 046**	0.018*	** 7700	****UU	0.049**	0.014**	**070 0
COUNCIL DI CERCETOR ACON	(0.020)	(0.053)	(0.019)	(0.033)	(0.062)	(0.024)	(0000)	(0.015)	(0.024)	(0.050)
Incumbent effect \mid joint pre-election years (E)	0.028^{*} (0.083)	0.038^{**} (0.042)	0.033* (0.066)	0.038^{*} (0.052)	0.032^{*} (0.064)	0.038^{**} (0.049)	0.035** (0.028)	0.039^{**} (0.028)	0.036* (0.073)	0.041^{*} (0.066)
Difference if incumbent reruns given joint elections Mayoral pre-election year $(D-C)$	0.016 (0.199)	0.050^{*} (0.051)	0.021 (0.129)	0.056^{**} (0.040)	0.020 (0.207)	0.055^{**} (0.033)	0.023^{*} (0.061)	0.056^{**} (0.028)	$0.022 \\ (0.114)$	0.057^{**} (0.039)
Council election year $(G-F)$		0.033 (0.167)		0.049^{**} (0.036)		0.044^{**} (0.035)		0.036^{*} (0.083)		0.046^{*} (0.051)
Council post-election year $(L-K)$		-0.069* (0.052)		-0.076^{**} (0.036)		-0.081^{**} (0.026)		-0.067** (0.040)		-0.054 (0.370)
Covariates included	~	~	~	~	~	~	~	~	~	~
Observations	41354	41354	38861	38861	39344	39344	39344	39344	39425	39425
R^2	0.079	0.078	0.076	0.075	0.076	0.075	0.080	0.079	0.076	0.076
Cluster	3011	3011	2963	2963	140	140	2997	2997	2997	2997
Underidentification test (Kleibergen-Paap)	870.0	729.8	814.3	673.8	92.2	89.9	833.2	674.8	303.6 20	187.8
weak identification test (Aleibergen-Faap) Overidentification test (Hansen)	0.957	$03.2 \\ 0.742$	$140.2 \\ 0.963$	$^{49.6}_{0.748}$	92.9 0.976	$^{4.3.9}_{0.601}$	0.982	$^{49.3}_{0.627}$	י מאיש	
Notes: The letters in italic font refer to the correspc completed election periods in our regression. In colu level instead of the municipal level. In columns (7) (9) and (10) , we only use the dummy indicating whi instrument for the instrumental variables regression.	nding letters mns (3) and (and (8), we n ether the may . Two-sided p-	in Table 4. The s 4), county-free cit replace the depen or is older than 6 values are shown	iensitivity ana dies are exclud dent variable 00 years in the in paranthese	ysis is done for ad from the estii with the log of mayoral electio s. ***, (**), (*)	columns (5) a mation sample municipal gro on year (and t denotes signi	nd (6) of Table . In columns (5) se expenditures, he interactions c ficance at the 1-	3. In column) and (6) , we i.e., we add of the instrum , $(5-)$, $(10-)\%$	s (1) and (2) cluster the e state transf nent with th rlevel. Sourc), we have ir srror term of ers. Finally, e election ir ce: Own cal	cluded non- n the county in columns (dicators) as culations.

•	regression
	variables
	Instrumental
-	test (
•	sensitivity
	penditure:
	public ex
•	S IN
()
F	η
ŕ	Ъ,
1	້
-	Эľ
5	а Б
E	-

7. Conclusion

In this paper, we study PBCs in expenditures with a particular focus on overlapping cycles in the executive and the legislative branch of local governments. As the literature on PBC usually cannot separate effects for these two political spheres, this is a distinct feature of our study. Furthermore, we directly account for the potential endogeneity of the incumbent's decision to re-run for office and how this decision affects spending in times of elections. The testing ground is given by German local elections for mayors and municipal councils in the two southern states of *Bavaria* (BAY) and *Baden-Wuerttemberg* (BW). For both states, we observe the universe of all municipalities during the period from 1992-2006.

Our main findings can be summarised as follows: 1) We find sizable and statistically significant unconditional pre-election effects for council elections. Total expenditures are increased between 1.3–1.8 percent in the year before the council election takes place. 2) These budget cycle effects for the legislative branch (council) are robust to the inclusion of election variables for the executive branch (mayor). 3) We find no average unconditional (pre-)election PBC for the mayoral elections. 4) This also holds for the interaction effects when we condition on whether the incumbent mayor runs for re-election. However, if we look at differences for the incumbent effect, 5) we find increased spending in pre-election years if the incumbent re-runs. Finally, 6) we find insightful differences for overlapping cycles conditional on the incumbent's decision to re-run for office. Municipal expenditures increase in joint pre-election and election years if the mayor runs for re-election and decrease in joint post-election years if she did not. These results are all confirmed by estimations using an instrumental variables approach for the potentially endogenous decision of the incumbent to seek re-election.

Our results support the following conclusion: First, joint elections of councils and mayors seem to matter for the formation of PBCs. Separating the local elections can thus lower politically motivated distortions of expenditures before elections. Second, our findings for increased spending before mayoral elections when the incumbent mayor re-runs may justify age and term limits. In BAY and BW, we have (in)direct age limits but no term limit restrictions. Abolishing the age limits would likely increase the number of incumbent candidates re-running and introducing term limits would act in the opposite direction. Nonetheless, we would refrain from drawing direct inferences from our results. On the one hand, strong age and term limit rules could be viewed as limiting overspending before joint mayoral and council elections. In contrast, however, the mayor's incentive to manipulate the single election where she could be re-elected (for instance, if we assume a two-term limit) should even increase. It is thus not obvious whether the introduction of term limits would decrease or increase the overall effect of strategic spending in the long run. This question is thus open for further research.

References

- Ade, F., Freier, R., 2013. Divided government versus incumbency externality effect quasiexperimental evidence on multiple voting decisions. European Economic Review 64, 1–20.
- Aidt, T. S., Veiga, F. J., Veiga, L. G., 2011. Election results and opportunistic policies: A new test of the rational political business cycle model. Public Choice 148 (1-2), 21–44.
- Akhmedov, A., Zhuravskaya, E., 2004. Opportunistic political cycles: Test in a young democracy setting. The Quarterly Journal of Economics 119 (4), 1301–1338.
- Alesina, A., 1987. Macroeconomic policy in a two-party system as a repeated game. The Quarterly Journal of Economics 102 (3), 651–678.
- Alesina, A., 1988a. Credibility and political convergence in a two-party system with rational voters. The American Economic Review 78 (4), 796–805.
- Alesina, A., 1988b. Macroeconomics and politics. National Bureau of Economic Research Macroeconomics Annual 3, 13–61.
- Alesina, A., Cohen, G. D., Roubini, N., 1992. Macroeconomic policies and elections in OECD democracies. Economics and Politics 4 (1), 1–30.
- Alesina, A., Roubini, N., Cohen, G. D., 1997. Political cycles and the macroeconomy. MIT Press.
- Alt, J. E., Rose, S. S., 2009. Context-conditional political cycles. In: Boix, C., Stokes, S. C. (Eds.), The Oxford Handbook of Comparative Politics. Oxford University Press, pp. 1–18.
- Angrist, J. D., Pischke, J.-S., 2009. Mostly harmless econometris: an empiricist's companion. Princeton University Press.
- Baleiras, R. N., da Silva Costa, J., 2004. To be or not to be in office again: an empirical test of a local political business cycle rational. European Journal of Political Economy 20 (3), 655–671.
- Benito, B., Bastida, F., Vicente, C., 2013. Creating room for manoeuvre: a strategy to generate political budget cycles under fiscal rules. Kyklos 66 (4), 467–496.
- Binet, M.-E., Pentecôte, J.-S., 2004. Tax degression and the political budget cycle in French municipalities. Applied Economics Letters 11 (14), 905–908.
- Blais, A., Nadeau, R., 1992. The electoral budget cycle. Public Choice 74 (4), 389–403.

- Brender, A., Drazen, A., 2005. Political budget cycles in new versus established economies. Journal of Monetary Economics 52 (7), 1271–1295.
- Cioffi, M., Messina, G., Tommasino, P., 2012. Parties, institutions and political budget cycles at municipal level: evidence from Italy. Banca d'Italia Working Paper 885.
- Drazen, A., Eslava, M., 2010. Electoral manipulation via expenditure composition: theory and evidence. Journal of Development Economics 92 (1), 39–52.
- Englmaier, F., Stowasser, T., 2013. Electoral cycles in savings bank lending. CESifo Working Paper No. 4402.
- Federal Statistical Office, 2011. Finanzen und Steuern: Rechnungsergebnisse der öffentlichen Haushalte. Federal Statistical Office, Wiesbaden.
- Foremny, D., Riedel, N., 2014. Business taxes and the electoral cycle. Journal of Public Economics 115, 48–61.
- Foucault, M., Madies, T., Paty, S., 2008. Public spending interactions and local politics. Empirical evidence from French municipalities. Public Choice 137, 57–80.
- Freier, R., 2011. Incumbency as the major advantage The scopes and determinants of the electoral advantage for incumbent mayors. DIW Working Paper Series No. 1177.
- Furdas, M., Homolkova, K., Kis-Katos, K., 2014. Local political budget cycles in a federation: Evidence from West German cities. Manuscript University of Freiburg.
- Galli, E., Rossi, S. P., 2002. Political budget cycle: The case of West German Länder. Public Choice 110 (3-4), 283–303.
- Golden, D. G., Poterba, J. M., 1980. The price of popularity: the political business cycle reexamined. American Journal of Political Science 24 (4), 696–714.
- Heinemann, F., Janeba, E., Moessinger, M.-D., Schröder, C., 2013. Revenue autonomy preference in German state parliaments. ZEW Discussion Paper No. 13-090.
- Hibbs, D. A. J., 1977. Political parties and macroeconomic policy. American Political Science Review 71 (4), 1467–1487.
- Keech, W. R., Pak, K., 1989. Electoral cycles and budgetary growth in veterans' benefit programs. American Journal of Political Science 33 (4), 901–911.
- Kern, T., 2008. Warum werden Bürgermeister abgewählt? Eine Studie aus Baden-Württemberg über den Zeitraum 1973 bis 2003. W. Kohlhammer, Stuttgart.

- Khemani, S., 2004. Political cycles in a developing economy: effect of elections in the Indian states. Journal of Development Economics 73, 125–154.
- Klomp, J., De Haan, J., 2013. Do political budget cycles really exist? Applied Economics 45, 329–341.
- Kneebone, R. D., Mckenzie, K. J., 2001. Electoral and partial cycles in fiscal policy: an examination of Canadian provinces. International Tax and Public Finance 8 (5-6), 753– 774.
- Lächler, U., 1978. The political business cycle: a complementary study. The Review of Economic Studies 45 (2), 369–375.
- Mechtel, M., Potrafke, N., 2013. Electoral cycles in active labour market policies. Public Choice 156 (1-2), 181–194.
- Nordhaus, W. D., 1975. The political business cycle. The Review of Economic Studies 42, 169–190.
- Persson, T., Tabellini, G., 2000. Political economics explaining economic policy. The MIT Press.
- Potrafke, N., 2012. Political cycles and economic performance in OECD countries: empirical evidence from 1951-2006. Public Choice 150, 155–179.
- Rogoff, K., 1990. Equilibrium political budget cycles. American Economic Review 80(1), 21–36.
- Rogoff, K., Sibert, A., 1988. Elections and macroeconomic policy cycles. Review of Economic Studies 55(1), 1–16.
- Rose, S. S., 2006. Do fiscal rules dampen the political business cycle. Public Choice 128 (3-4), 407–431.
- Rosenberg, J., 1992. Rationality and the political business cycle: The case of local government. Public Choice 73 (1), 71–81.
- Schneider, C. J., 2010. Fighting with one hand tied behind the back: political budget cycles in the West German states. Public Choice 142, 125–150.
- Schuknecht, L., 1999. Fiscal policy cycles and the exchange rate regime in developing countries. European Journal of Political Economy 15 (3), 569–580.
- Seitz, H., 2000. Fiscal policy, deficits and politics of subnational governments: The case of German Länder. Public Choice 102 (3-4), 183–218.

- Shi, M., Svensson, J., 2006. Political budget cycles: Do they differ across countries and why? Journal of Public Economics 90, 1367–1389.
- Veiga, L. G., Veiga, F. J., 2007a. Does opportunism pay off? Economics Letters 96 (2), 177–182.
- Veiga, L. G., Veiga, F. J., 2007b. Political business cycle at the municipal level. Public Choice 131 (1-2), 45–64.

Wooldridge, J. M., 2010. Econometric analysis of cross section and panel data. MIT Press.

A. Appendix

Study	Dependent variable	Main results	Country	Level of gov- ernment	Political	Variation
Aidt et al. (2011)	Deviation from term-mean of investment expenditure, total expenditure, current expenditure and capital expenditure in years of election; as well as their	Lower average expenditures during the term lead to higher opportunistic deviations in election years, effect intensified with decreasing vote- margins; In case of absolute values, expenditure in election years depends positively on those in pre-election years	Portugal	Municipal	Combined local elections	Time variation
Akhmedov and Zhuravskaya (2004)	absolute values Total expenditure, composition of expenditure, growth,	Increased expenditure, expenditure shifts to visi- ble expenditures 1-2 months before elections, mag- nitude of effects decrease with increasing institu- tional duality.	Russia	Regional	Regional governor elections	Time variation
Alt and Rose (2009)	Total expenditure	Increased expenditure, cycles intensified in case of Increased expenditure, cycles intensified in case of medium approval ratings for the incumbent, uni- fied government, low level of transparency, low level of media menetration lacking facel miles	USA	State	Gubernatorial elections	Time variation
Baleiras and da Silva Costa (2004)	Investment expenditure	Increased expenditure in pre-election years	Portugal	Municipal	Combined local elections	Time variation
(2007) Blais and Nadeau (1992)	Total expenditure, surplus/deficit, composition of	Increased total expenditure in election years, ex- penditure shifted to social services and roads, in- creased deficits in election years	Canada	Provincial	Provincial elections	Time variation
Benito et al. (2013)	expending Budget deficits, capital spending, revenues	Slight reduction or no change in deficits, stronger reduction if fiscal rule prevalent, increase in rev- enues before and after elections, increase in ex- penditure only in year before elections, reduction in available only in semiclone in environment	Spain	Municipal	Municipal legislative elections	Time variation
Binet and Pentecôte (2004)	Per capita tax revenue, per capita capital expenditure, per capita	Decreased tax revenues, increased capital expen- ditures, increased loans in pre-election years	France	Municipal	Municipal council elections	Time variation
Cioffi et al. (2012)	Per capita total expenditure, per capita capital expenditure	Expenditures increase during the years of the elec- tion and pre-election years. Cycles are more pro- nounced if the mayor is not affiliated to a national	Italy	Municipal	Municipal council elections	Time variation
Drazen and Eslava (2010)	Composition of expenditure	party. Expenditure shifted to more targeted categories such as health, infrastructure spending or housing	Colombia	Municipal	Mayor and city-council	Time varia- tion

Table A.1: Studies on political budget cycles in expenditures for sub-national levels of government

Continued on next page.

Study	Dependent variable	Main results	Country	Level of gov- ernment	Political institution	Variation elections
Foucault et al. (2008)	Primary expenditure, operating expenditure, payroll expenditure, investment	Increased expenditures in all types tested in year before election, expenditures decrease in year af- ter the election, effects strongest for investment expenditure	France	Municipal	Municipal council elections	Time variation
Galli and Rossi (2002)	Total action of composition of compo	Increased expenditure, increased deficits in elec- tion years, no clear shifts in expenditure, no ideo- logical cycles	Germany	State	State elections	Time variation
Khemani (2004)	Tax revenues, own revenues, various expenditure categories, budget deficit	No effect of fiscal expansions on overall spending and budget deficits but small manipulation of tar- geted expenditures, i.e., there is higher investment spending while spending in populist regarded cat- egories is lower. Furthermore, there is lower tax	India	State	State elections	Time variation
Kneebone and Mckenzie (2001)	Primary deficit, total expenditure, revenue, commosition of exm	Increased expenditure, increased deficits in elec- tion years, expenditure shifted to schools (effects depend on ideology)	Canada	Provincial	Provincial elections	Time variation
Rose (2006)	Budget deficits, total expenditure, revenues	Deficit increases in election year, weaker increase if fiscal rules prevalent, taxes unaffected, expendi- tures increase in election years, increase stronger under unified government, effects weakened if fis- col vib is memory.	USA	State	Gubernatorial elections	Time variation
Rosenberg (1992)	Development expenditures	Increased expenditure in election years, effect in- tensified if incumbent does not re-run	Israel	City	City-council elections	Between variation
Schneider (2010)	Deficit, total expenditure, composition of	Decline in deficits, no changes in expenditure, composition of expenditure, expenditure shifted to	Germany	State	State elections	Time variation
Seitz (2000)	Total expenditure, composition of expenditure, primary	No partisan impact on total expenditure and its components	Germany	State	State elections	Time variation
Veiga and Veiga (2007b)	Budget balance, taxes, total expenditure, composition of expenditure	Increased deficits, reduced taxes, increased total expenditures in election years, effects intensified for left-wing mayors	Portugal	Municipal	Combined local elections	Time varia- tion
Notes: The summar	y comprises studies that are ei	ther published in refereed journals or as working pape	ers. All pape	ers are cited in ou	ır references.	

Table A.1: continued

Variable	Description	Source
Total expenditures in logs (net of transfers) Dummy pre-election year	Logarithm of total municipal expenditures minus state transfers. Dummy $= 1$ if there is a council election, state election, or	Jahresrechnungsstatistik Statistical office (BAY), Onortionairo internot meeorol (BWI)
Dummy election year	Dummy = 1 if there is a council election, state election, or	Statistical office (BAY),
Dummy post-election year	mayor election in the current year; zero otherwise. Dummy $= 1$ if there is a council election, state election, or	Questionaire, internet research (BW) Statistical office (BAY),
Dummy incumbent runs again	mayor election in the preceding year; zero otherwise. Dummy $= 1$ if the incumbent mayor runs for re-election; zero otherwise.	Questionaire, internet research (BW) Statistical office (BAY),
Dummy full time mayor	The variable is equal to $one/zero$ for the whole election period. Dummy = 1 if the mayor is working as a full time mayor; zero otherwise.	Questionaire, internet research (BW) Statistical office (BAY),
Dummy left mayor	Dummy $= 1$ if the mayor is member of the Social Democrats (SPD) or	Questionaire, internet research (BW) Statistical office (BAY),
Dummy mayor is pensionable	the Green party (Grüne); zero otherwise. Dummy $= 1$ if the mayor is eligible to receive pensions: zero otherwise.	Internet research (BW) Own computation based on
-	The variable is equal to one/zero for the whole election period.	state regulations
Dummy mayor is older than 60 years	Dummy = 1 if the mayor is 60 years old or older in the election year; zero otherwise. The variable is equal to $one/zero$ for the whole election period.	Own computation
Age of the mayor	Age of the mayor in years.	Statistical office (BAY), Internet research (BW)
Share left parties in council	Number of seats of the Social Democrats (SPD) and the Green party (Grüne) divided by the total number of seats.	Statistical offices (BAY and BW)
Population (per thousand)	Municipal population divided by 1,000.	Statistical offices (BAY and BW)
Population (per thousand) squared	Squared population (per thousand).	Statistical offices (BAY and BW)
Share population < 15 years	Number of inhabitants younger than 15 years divided by total population.	Statistical offices (BAY and BW)
Share population ≥ 65 years	Number of inhabitants older than 65 years divided by total population.	Statistical offices (BAY and BW)

Table A.2: Variable description

Table A.3: Sample selection comparison for BW

	Full San	nple	Estimation	Sample	t-test mean comparison
	No/Mean	S.D.	No/Mean	S.D.	Hypothesis: Difference $\neq 0$
Number of municipalities	1101	-	968	-	-
Municipal-year observations	16515	-	10531	-	-
Log total exp. (net of transfers)	15.964	1.182	16.116	1.143	$\Pr(T > t) = 0.0000$
Population (per thousands)	9.481	25.54	10.800	28.19	$\Pr(T > t) = 0.0001$
Population share < 15	0.182	0.024	0.180	0.024	$\Pr(T > t) = 0.0000$
Population share > 65	0.151	0.029	0.155	0.029	$\Pr(T > t) = 0.0000$

Notes: We test against the null hypothesis that the difference in means is equal to zero, i.e., H0: mean(full sample) - mean(estimation sample) = 0. The p-values show that the difference is not equal to zero.

	Depe (1)	endent variabl (2)	e: total exper (3)	nditures in log (4)	s (net of trans (5)	sfers) (6)
Share left parties in council	$0.024 \\ [0.571]$	$0.024 \\ [0.580]$	0.023 [0.554]	0.023 [0.554]	0.023 [0.555]	0.023 [0.552]
Dummy left mayor	-0.008 $[-0.774]$	-0.008 $[-0.771]$	-0.008 [-0.766]	-0.008 $[-0.765]$	-0.008 [-0.762]	-0.008 $[-0.754]$
Dummy full time mayor	0.012 [0.815]	0.012 [0.797]	0.012 [0.807]	0.012 [0.805]	0.012 [0.809]	0.012 [0.800]
Age of the mayor	0.000 [0.700]	0.000 [0.863]	0.000 [0.836]	0.000 [0.840]	0.000 [0.133]	$0.000 \\ [0.140]$
State elections						
Pre-election year	-0.003 [-0.527]	-0.004 [-1.066]	-0.002 [-0.284]	$0.001 \\ [0.091]$	-0.002 [-0.341]	0.000 [0.048]
Election year	0.011^{**} [2.088]	$0.003 \\ [1.075]$	0.010^{*} $[1.799]$	0.009^{*} [1.746]	0.010^{*} $[1.856]$	0.010^{*} $[1.854]$
Post-election year	0.021^{***} [4.028]	0.024^{***} [7.054]	0.021^{***} [4.025]	0.023^{***} [4.324]	0.021^{***} [3.952]	0.023^{***} [4.307]
Population	0.018^{***} [3.377]	0.018^{***} [3.418]	0.018^{***} [3.402]	0.018^{***} [3.409]	0.018^{***} [3.404]	0.018^{***} [3.407]
Population squared	-0.007*** [-2.993]	-0.007^{***} [-3.037]	-0.007*** [-3.029]	-0.007^{***} [-3.037]	-0.007^{***} [-3.014]	-0.007*** [-3.029]
Share population < 15 years	0.288 [1.156]	0.293 [1.177]	0.288 $[1.155]$	0.288 [1.158]	$0.290 \\ [1.164]$	$0.292 \\ [1.171]$
Share population > 65 years	$0.131 \\ [0.737]$	$0.181 \\ [1.096]$	$0.142 \\ [0.792]$	$0.151 \\ [0.831]$	$0.151 \\ [0.842]$	$0.160 \\ [0.878]$
Trend BAY	0.029^{***} [14.338]	0.020^{***} [9.572]	0.030^{***} [14.364]	0.030^{***} [14.358]	0.030^{***} [14.421]	0.030^{***} [14.407]
Trend BAY squared	-0.001*** [-5.980]	-0.000*** [-2.977]	-0.001*** [-6.018]	-0.001^{***} [-6.015]	-0.001^{***} [-6.079]	-0.001*** [-6.077]
Trend BW	0.036^{***} [13.353]	0.025^{***} [9.853]	0.036^{***} $[13.396]$	0.036^{***} [13.344]	0.036^{***} [13.352]	0.036^{***} [13.345]
Trend BW squared	-0.001*** [-5.836]	-0.000* [-1.762]	-0.001*** [-5.912]	-0.001*** [-5.888]	-0.001*** [-5.923]	-0.001*** [-5.923]
Constant	15.153^{***} [226.984]	15.141^{***} [229.019]	15.147^{***} [225.610]	15.146^{***} [224.519]	15.165^{***} [219.643]	15.163^{***} [218.392]
Observations R^2 Cluster	$39554 \\ 0.077 \\ 3017$	$39554 \\ 0.077 \\ 3017$	$39554 \\ 0.077 \\ 3017$	$39554 \\ 0.077 \\ 3017$	$39554 \\ 0.077 \\ 3017$	$39554 \\ 0.077 \\ 3017$

Table A.4: PBCs in public expenditure: Control variables for Table 2

Notes: All specifications include municipality and time fixed effects. Robust t-values are shown in brackets. ***, (**), (*) denotes significance at the 1-, (5-), (10-)%-level. Error terms are clustered on the municipal level.

	Depende (1)	ent variable: (2)	$\begin{array}{c} \text{Dummy} = 1 \\ (3) \end{array}$	if the incumb (4)	ent runs for r (5)	e-election (6)
Instrument 1: Dummy mayor is pensionable	-0.196*** [-15.249]	-0.194*** [-15.016]	-0.193*** [-14.993]	-0.193 ^{***} [-14.976]	-0.260 ^{***} [-17.160]	-0.275*** [-17.086]
Instrument 2: Dummy mayor is older than 60 years	-0.358*** [-26.568]	-0.354*** [-26.258]	-0.354*** [-26.243]	-0.354*** [-26.240]	-0.421^{***} [-25.577]	-0.393*** [-22.991]
Council elections Pre-election year	0.011		0.021**	0.001	0.029***	-0.014
Election year	0.001		0.034***	-0.006	0.043***	-0.026**
Post-election year	0.015**		0.041***	-0.001	0.064***	-0.004
Mayor elections Pre-election year	[2.030]	0.006	-0.000	-0.011	-0.020***	-0.047***
Election year		-0.054***	[-0.076] -0.061***	[-1.626] -0.096***	[-3.821] -0.077***	[-5.839] -0.138***
Post-election year		[-8.369] -0.042***	[-8.696] -0.054^{***}	[-9.124] -0.089***	[-9.824] -0.416***	[-10.386] -0.585^{***}
Mayor elections × council elections		[-4.617]	[-5.129]	[-5.916] 0.031*	[-32.439]	[-25.287]
Joint pre-election year				[1.788]		[3.136]
Joint election year				0.092^{***} [4.825]		0.153^{***} [7.017]
Joint post-election year				0.101^{***} [4.378]		0.287^{***} [10.770]
Pre-election year					0.022^{***} [2.917]	0.091^{***} [6.168]
Election year					0.039^{***} [3.603]	0.154^{***} [7.258]
Post-election year					0.480^{***}	0.695^{***} [25-584]
Mayor elections \times instrument 2 Pre-election year					0.016* [1.912]	-0.076*** [-3.157]
Election year					-0.020 [-1.544]	-0.216*** [-7.094]
Post-election year					0.465***	0.489***
Council elections \times instrument 1 Pre-election year					[23.333]	0.086*** [5 415]
Election year						0.090***
Post-election year						0.081***
Council elections \times instrument 2 Pre-election year						-0.102***
Election year						[-3.958] -0.152***
Post-election year						-0.148***
Joint elections \times instrument 1 Pre-election years						[-6.689] -0.182^{***}
Election years						[-5.700] -0.256^{***}
Part election many						[-7.422]
Joint elections \times instrument 2						[-12.668]
Pre-election years						0.215^{***} [4.644]
Election years						0.385^{***} [8.401]
Post-election years						0.136^{***} [4.330]
Control variables	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
Observations R^2	$39363 \\ 0.220$	$39363 \\ 0.222$	$39363 \\ 0.222$	$39363 \\ 0.223$	$39363 \\ 0.335$	$39363 \\ 0.343$

Table A.5: PBCs in public expenditure: First stage regression of Table 3

Notes: All specifications include municipality and time fixed effects. Robust t-values are shown in brackets. ***, (**), (*) denotes significance at the 1-, (5-), (10-)%-level. Error terms are clustered on the municipal level. The estimation includes the full set of additional control variables. Source: Own calculations.

2011

2011/1, Oppedisano, V; Turati, G.: "What are the causes of educational inequalities and of their evolution over time in Europe? Evidence from PISA"

2011/2, Dahlberg, M; Edmark, K; Lundqvist, H.: "Ethnic diversity and preferences for redistribution"

2011/3, Canova, L.; Vaglio, A.: "Why do educated mothers matter? A model of parental help"

2011/4, Delgado, F.J.; Lago-Peñas, S.; Mayor, M.: "On the determinants of local tax rates: new evidence from Spain"

2011/5, Piolatto, A.; Schuett, F.: "A model of music piracy with popularity-dependent copying costs"

2011/6, Duch, N.; García-Estévez, J.; Parellada, M.: "Universities and regional economic growth in Spanish regions"

2011/7, Duch, N.; García-Estévez, J.: "Do universities affect firms' location decisions? Evidence from Spain"

2011/8, Dahlberg, M.; Mörk, E.: "Is there an election cycle in public employment? Separating time effects from election year effects"

2011/9, Costas-Pérez, E.; Solé-Ollé, A.; Sorribas-Navarro, P.: "Corruption scandals, press reporting, and accountability. Evidence from Spanish mayors"

2011/10, Choi, A.; Calero, J.; Escardíbul, J.O.: "Hell to touch the sky? Private tutoring and academic achievement in Korea"

2011/11, Mira Godinho, M.; Cartaxo, R.: "University patenting, licensing and technology transfer: how organizational context and available resources determine performance"

2011/12, Duch-Brown, N.; García-Quevedo, J.; Montolio, D.: "The link between public support and private R&D effort: What is the optimal subsidy?"

2011/13, Breuillé, M.L.; Duran-Vigneron, P.; Samson, A.L.: "To assemble to resemble? A study of tax disparities among French municipalities"

2011/14, McCann, P.; Ortega-Argilés, R.: "Smart specialisation, regional growth and applications to EU cohesion policy"

2011/15, Montolio, D.; Trillas, F.: "Regulatory federalism and industrial policy in broadband telecommunications"

2011/16, Pelegrín, A.; Bolancé, C.: "Offshoring and company characteristics: some evidence from the analysis of Spanish firm data"

2011/17, Lin, C.: "Give me your wired and your highly skilled: measuring the impact of immigration policy on employers and shareholders"

2011/18, **Bianchini**, **L.**; **Revelli**, **F.**: "Green polities: urban environmental performance and government popularity" **2011/19**, **López Real**, **J.**: "Family reunification or point-based immigration system? The case of the U.S. and Mexico"

2011/20, Bogliacino, F.; Piva, M.; Vivarelli, M.: "The impact of R&D on employment in Europe: a firm-level analysis"

2011/21, Tonello, M.: "Mechanisms of peer interactions between native and non-native students: rejection or integration?"

2011/22, García-Quevedo, J.; Mas-Verdú, F.; Montolio, D.: "What type of innovative firms acquire knowledge intensive services and from which suppliers?"

2011/23, Banal-Estañol, A.; Macho-Stadler, I.; Pérez-Castrillo, D.: "Research output from university-industry collaborative projects"

2011/24, Ligthart, J.E.; Van Oudheusden, P.: "In government we trust: the role of fiscal decentralization"

2011/25, Mongrain, S.; Wilson, J.D.: "Tax competition with heterogeneous capital mobility"

2011/26, Caruso, R.; Costa, J.; Ricciuti, R.: "The probability of military rule in Africa, 1970-2007"

2011/27, Solé-Ollé, A.; Viladecans-Marsal, E.: "Local spending and the housing boom"

2011/28, Simón, H.; Ramos, R.; Sanromá, E.: "Occupational mobility of immigrants in a low skilled economy. The Spanish case"

2011/29, Piolatto, A.; Trotin, G.: "Optimal tax enforcement under prospect theory"

2011/30, Montolio, D; Piolatto, A.: "Financing public education when altruistic agents have retirement concerns"

2011/31, García-Quevedo, J.; Pellegrino, G.; Vivarelli, M.: "The determinants of YICs' R&D activity"

2011/32, Goodspeed, T.J.: "Corruption, accountability, and decentralization: theory and evidence from Mexico"

2011/33, **Pedraja**, **F.**; **Cordero**, **J.M.**: "Analysis of alternative proposals to reform the Spanish intergovernmental transfer system for municipalities"

2011/34, Jofre-Monseny, J.; Sorribas-Navarro, P.; Vázquez-Grenno, J.: "Welfare spending and ethnic heterogeneity: evidence from a massive immigration wave"

2011/35, Lyytikäinen, T.: "Tax competition among local governments: evidence from a property tax reform in Finland"

2011/36, Brülhart, M.; Schmidheiny, K.: "Estimating the Rivalness of State-Level Inward FDI"

2011/37, García-Pérez, J.I.; Hidalgo-Hidalgo, M.; Robles-Zurita, J.A.: "Does grade retention affect achievement? Some evidence from Pisa"

2011/38, Boffa, f.; Panzar. J.: "Bottleneck co-ownership as a regulatory alternative"

2011/39, González-Val, R.; Olmo, J.: "Growth in a cross-section of cities: location, increasing returns or random growth?"

2011/40, Anesi, V.; De Donder, P.: "Voting under the threat of secession: accommodation vs. repression"

2011/41, Di Pietro, G.; Mora, T.: "The effect of the l'Aquila earthquake on labour market outcomes"

2011/42, Brueckner, J.K.; Neumark, D.: "Beaches, sunshine, and public-sector pay: theory and evidence on amenities and rent extraction by government workers"

2011/43, Cortés, D.: "Decentralization of government and contracting with the private sector"

2011/44, Turati, G.; Montolio, D.; Piacenza, M.: "Fiscal decentralisation, private school funding, and students' achievements. A tale from two Roman catholic countries"

2012

2012/1, Montolio, D.; Trujillo, E.: "What drives investment in telecommunications? The role of regulation, firms' internationalization and market knowledge"

2012/2, Giesen, K.; Suedekum, J.: "The size distribution across all "cities": a unifying approach"

2012/3, Foremny, D.; Riedel, N.: "Business taxes and the electoral cycle"

2012/4, García-Estévez, J.; Duch-Brown, N.: "Student graduation: to what extent does university expenditure matter?"

2012/5, Durán-Cabré, J.M.; Esteller-Moré, A.; Salvadori, L.: "Empirical evidence on horizontal competition in tax enforcement"

2012/6, Pickering, A.C.; Rockey, J.: "Ideology and the growth of US state government"

2012/7, Vergolini, L.; Zanini, N.: "How does aid matter? The effect of financial aid on university enrolment decisions"

2012/8, Backus, P.: "Gibrat's law and legacy for non-profit organisations: a non-parametric analysis"

2012/9, Jofre-Monseny, J.; Marín-López, R.; Viladecans-Marsal, E.: "What underlies localization and urbanization economies? Evidence from the location of new firms"

2012/10, Mantovani, A.; Vandekerckhove, J.: "The strategic interplay between bundling and merging in complementary markets"

2012/11, Garcia-López, M.A.: "Urban spatial structure, suburbanization and transportation in Barcelona"

2012/12, Revelli, F.: "Business taxation and economic performance in hierarchical government structures"

2012/13, Arqué-Castells, P.; Mohnen, P.: "Sunk costs, extensive R&D subsidies and permanent inducement effects"

2012/14, Boffa, F.; Piolatto, A.; Ponzetto, G.: "Centralization and accountability: theory and evidence from the Clean Air Act"

2012/15, Cheshire, P.C.; Hilber, C.A.L.; Kaplanis, I.: "Land use regulation and productivity – land matters: evidence from a UK supermarket chain"

2012/16, Choi, A.; Calero, J.: "The contribution of the disabled to the attainment of the Europe 2020 strategy headline targets"

2012/17, Silva, J.I.; Vázquez-Grenno, J.: "The ins and outs of unemployment in a two-tier labor market"

2012/18, González-Val, R.; Lanaspa, L.; Sanz, F.: "New evidence on Gibrat's law for cities"

2012/19, Vázquez-Grenno, J.: "Job search methods in times of crisis: native and immigrant strategies in Spain"

2012/20, Lessmann, C.: "Regional inequality and decentralization – an empirical analysis"

2012/21, Nuevo-Chiquero, A.: "Trends in shotgun marriages: the pill, the will or the cost?"

2012/22, Piil Damm, A.: "Neighborhood quality and labor market outcomes: evidence from quasi-random neighborhood assignment of immigrants"

2012/23, Ploeckl, F.: "Space, settlements, towns: the influence of geography and market access on settlement distribution and urbanization"

2012/24, **Algan**, **Y**.; **Hémet**, **C**.; **Laitin**, **D**.: "Diversity and local public goods: a natural experiment with exogenous residential allocation"

2012/25, Martinez, D.; Sjögren, T.: "Vertical externalities with lump-sum taxes: how much difference does unemployment make?"

2012/26, Cubel, M.; Sanchez-Pages, S.: "The effect of within-group inequality in a conflict against a unitary threat"

2012/27, Andini, M.; De Blasio, G.; Duranton, G.; Strange, W.C.: "Marshallian labor market pooling: evidence from Italy"

2012/28, Solé-Ollé, A.; Viladecans-Marsal, E.: "Do political parties matter for local land use policies?"

2012/29, Buonanno, P.; Durante, R.; Prarolo, G.; Vanin, P.: "Poor institutions, rich mines: resource curse and the origins of the Sicilian mafia"

2012/30, Anghel, B.; Cabrales, A.; Carro, J.M.: "Evaluating a bilingual education program in Spain: the impact beyond foreign language learning"

2012/31, Curto-Grau, M.; Solé-Ollé, A.; Sorribas-Navarro, P.: "Partisan targeting of inter-governmental transfers & state interference in local elections: evidence from Spain"

2012/32, Kappeler, A.; Solé-Ollé, A.; Stephan, A.; Välilä, T.: "Does fiscal decentralization foster regional investment in productive infrastructure?"

2012/33, Rizzo, L.; Zanardi, A.: "Single vs double ballot and party coalitions: the impact on fiscal policy. Evidence from Italy"

2012/34, Ramachandran, R.: "Language use in education and primary schooling attainment: evidence from a natural experiment in Ethiopia"

2012/35, Rothstein, J.: "Teacher quality policy when supply matters"

2012/36, Ahlfeldt, G.M.: "The hidden dimensions of urbanity"

2012/37, Mora, T.; Gil, J.; Sicras-Mainar, A.: "The influence of BMI, obesity and overweight on medical costs: a panel data approach"

2012/38, Pelegrín, A.; García-Quevedo, J.: "Which firms are involved in foreign vertical integration?"

2012/39, Agasisti, T.; Longobardi, S.: "Inequality in education: can Italian disadvantaged students close the gap? A focus on resilience in the Italian school system"

2013

2013/1, Sánchez-Vidal, M.; González-Val, R.; Viladecans-Marsal, E.: "Sequential city growth in the US: does age matter?"

2013/2, Hortas Rico, M.: "Sprawl, blight and the role of urban containment policies. Evidence from US cities"

2013/3, Lampón, J.F.; Cabanelas-Lorenzo, P-; Lago-Peñas, S.: "Why firms relocate their production overseas? The answer lies inside: corporate, logistic and technological determinants"

2013/4, Montolio, D.; Planells, S.: "Does tourism boost criminal activity? Evidence from a top touristic country"

2013/5, Garcia-López, M.A.; Holl, A.; Viladecans-Marsal, E.: "Suburbanization and highways: when the Romans, the Bourbons and the first cars still shape Spanish cities"

2013/6, Bosch, N.; Espasa, M.; Montolio, D.: "Should large Spanish municipalities be financially compensated? Costs and benefits of being a capital/central municipality"

2013/7, Escardíbul, J.O.; Mora, T.: "Teacher gender and student performance in mathematics. Evidence from Catalonia"

2013/8, Arqué-Castells, P.; Viladecans-Marsal, E.: "Banking towards development: evidence from the Spanish banking expansion plan"

2013/9, Asensio, J.; Gómez-Lobo, A.; Matas, A.: "How effective are policies to reduce gasoline consumption? Evaluating a quasi-natural experiment in Spain"

2013/10, Jofre-Monseny, J.: "The effects of unemployment benefits on migration in lagging regions"

2013/11, Segarra, A.; García-Quevedo, J.; Teruel, M.: "Financial constraints and the failure of innovation projects"

2013/12, Jerrim, J.; Choi, A.: "The mathematics skills of school children: How does England compare to the high performing East Asian jurisdictions?"

2013/13, González-Val, R.; Tirado-Fabregat, D.A.; Viladecans-Marsal, E.: "Market potential and city growth: Spain 1860-1960"

2013/14, Lundqvist, H.: "Is it worth it? On the returns to holding political office"

2013/15, Ahlfeldt, G.M.; Maennig, W.: "Homevoters vs. leasevoters: a spatial analysis of airport effects"

2013/16, Lampón, J.F.; Lago-Peñas, S.: "Factors behind international relocation and changes in production

geography in the European automobile components industry"

2013/17, **Guío**, **J.M.**; **Choi**, **A.**: "Evolution of the school failure risk during the 2000 decade in Spain: analysis of Pisa results with a two-level logistic mode"

2013/18, Dahlby, B.; Rodden, J.: "A political economy model of the vertical fiscal gap and vertical fiscal imbalances in a federation"

2013/19, Acacia, F.; Cubel, M.: "Strategic voting and happiness"

2013/20, Hellerstein, J.K.; Kutzbach, M.J.; Neumark, D.: "Do labor market networks have an important spatial dimension?"

2013/21, Pellegrino, G.; Savona, M.: "Is money all? Financing versus knowledge and demand constraints to innovation"

2013/22, Lin, J.: "Regional resilience"

2013/23, Costa-Campi, M.T.; Duch-Brown, N.; García-Quevedo, J.: "R&D drivers and obstacles to innovation in the energy industry"

2013/24, Huisman, R.; Stradnic, V.; Westgaard, S.: "Renewable energy and electricity prices: indirect empirical evidence from hydro power"

2013/25, Dargaud, E.; Mantovani, A.; Reggiani, C.: "The fight against cartels: a transatlantic perspective"

2013/26, Lambertini, L.; Mantovani, A.: "Feedback equilibria in a dynamic renewable resource oligopoly: preemption, voracity and exhaustion" 2013/27, Feld, L.P.; Kalb, A.; Moessinger, M.D.; Osterloh, S.: "Sovereign bond market reactions to fiscal rules and no-bailout clauses – the Swiss experience"

2013/28, Hilber, C.A.L.; Vermeulen, W.: "The impact of supply constraints on house prices in England"

2013/29, Revelli, F .: "Tax limits and local democracy"

2013/30, Wang, R.; Wang, W.: "Dress-up contest: a dark side of fiscal decentralization"

2013/31, Dargaud, E.; Mantovani, A.; Reggiani, C.: "The fight against cartels: a transatlantic perspective"

2013/32, Saarimaa, T.; Tukiainen, J.: "Local representation and strategic voting: evidence from electoral boundary reforms"

2013/33, Agasisti, T.; Murtinu, S.: "Are we wasting public money? No! The effects of grants on Italian university students' performances"

2013/34, Flacher, D.; Harari-Kermadec, H.; Moulin, L.: "Financing higher education: a contributory scheme"

2013/35, Carozzi, F.; Repetto, L.: "Sending the pork home: birth town bias in transfers to Italian municipalities"

2013/36, Coad, A.; Frankish, J.S.; Roberts, R.G.; Storey, D.J.: "New venture survival and growth: Does the fog lift?"

2013/37, **Giulietti**, **M.**; **Grossi**, **L.**; **Waterson**, **M.**: "Revenues from storage in a competitive electricity market: Empirical evidence from Great Britain"

2014

2014/1, Montolio, D.; Planells-Struse, S.: "When police patrols matter. The effect of police proximity on citizens' crime risk perception"

2014/2, Garcia-López, M.A.; Solé-Ollé, A.; Viladecans-Marsal, E.: "Do land use policies follow road construction?"

2014/3, Piolatto, A.; Rablen, M.D.: "Prospect theory and tax evasion: a reconsideration of the Yitzhaki puzzle"

2014/4, Cuberes, D.; González-Val, R.: "The effect of the Spanish Reconquest on Iberian Cities"

2014/5, Durán-Cabré, J.M.; Esteller-Moré, E.: "Tax professionals' view of the Spanish tax system: efficiency, equity and tax planning"

2014/6, Cubel, M.; Sanchez-Pages, S.: "Difference-form group contests"

2014/7, Del Rey, E.; Racionero, M.: "Choosing the type of income-contingent loan: risk-sharing versus risk-pooling"

2014/8, Torregrosa Hetland, S.: "A fiscal revolution? Progressivity in the Spanish tax system, 1960-1990"

2014/9, Piolatto, A.: "Itemised deductions: a device to reduce tax evasion"

2014/10, Costa, M.T.; García-Quevedo, J.; Segarra, A.: "Energy efficiency determinants: an empirical analysis of Spanish innovative firms"

2014/11, García-Quevedo, J.; Pellegrino, G.; Savona, M.: "Reviving demand-pull perspectives: the effect of demand uncertainty and stagnancy on R&D strategy"

2014/12, Calero, J.; Escardíbul, J.O.: "Barriers to non-formal professional training in Spain in periods of economic growth and crisis. An analysis with special attention to the effect of the previous human capital of workers"

2014/13, Cubel, M.; Sanchez-Pages, S.: "Gender differences and stereotypes in the beauty"

2014/14, Piolatto, A.; Schuett, F.: "Media competition and electoral politics"

2014/15, Montolio, D.; Trillas, F.; Trujillo-Baute, E.: "Regulatory environment and firm performance in EU telecommunications services"

2014/16, **Lopez-Rodriguez**, **J.**; **Martinez**, **D.**: "Beyond the R&D effects on innovation: the contribution of non-R&D activities to TFP growth in the EU"

2014/17, González-Val, R.: "Cross-sectional growth in US cities from 1990 to 2000"

2014/18, Vona, F.; Nicolli, F.: "Energy market liberalization and renewable energy policies in OECD countries"

2014/19, Curto-Grau, M.: "Voters' responsiveness to public employment policies"

2014/20, Duro, J.A.; Teixidó-Figueras, J.; Padilla, E.: "The causal factors of international inequality in co2 emissions per capita: a regression-based inequality decomposition analysis"

2014/21, Fleten, S.E.; Huisman, R.; Kilic, M.; Pennings, E.; Westgaard, S.: "Electricity futures prices: time varying sensitivity to fundamentals"

2014/22, Afcha, S.; García-Quevedo, J.: "The impact of R&D subsidies on R&D employment composition"

2014/23, Mir-Artigues, P.; del Río, P.: "Combining tariffs, investment subsidies and soft loans in a renewable electricity deployment policy"

2014/24, Romero-Jordán, D.; del Río, P.; Peñasco, C.: "Household electricity demand in Spanish regions. Public policy implications"

2014/25, Salinas, P.: "The effect of decentralization on educational outcomes: real autonomy matters!"

2014/26, Solé-Ollé, A.; Sorribas-Navarro, P.: "Does corruption erode trust in government? Evidence from a recent surge of local scandals in Spain"

2014/27, Costas-Pérez, E.: "Political corruption and voter turnout: mobilization or disaffection?"

2014/28, Cubel, M.; Nuevo-Chiquero, A.; Sanchez-Pages, S.; Vidal-Fernandez, M.: "Do personality traits affect productivity? Evidence from the LAB"

2014/29, Teresa Costa, M.T.; Trujillo-Baute, E.: "Retail price effects of feed-in tariff regulation"

2014/30, Kilic, M.; Trujillo-Baute, E.: "The stabilizing effect of hydro reservoir levels on intraday power prices under wind forecast errors"

2014/31, Costa-Campi, M.T.; Duch-Brown, N.: "The diffusion of patented oil and gas technology with environmental uses: a forward patent citation analysis"

2014/32, Ramos, R.; Sanromá, E.; Simón, H.: "Public-private sector wage differentials by type of contract: evidence from Spain"

2014/33, Backus, P.; Esteller-Moré, A.: "Is income redistribution a form of insurance, a public good or both?"

2014/34, Huisman, R.; Trujillo-Baute, E.: "Costs of power supply flexibility: the indirect impact of a Spanish policy change"

2014/35, Jerrim, J.; Choi, A.; Simancas Rodríguez, R.: "Two-sample two-stage least squares (TSTSLS) estimates of earnings mobility: how consistent are they?"

2014/36, Mantovani, A.; Tarola, O.; Vergari, C.: "Hedonic quality, social norms, and environmental campaigns" 2014/37, Ferraresi, M.; Galmarini, U.; Rizzo, L.: "Local infrastructures and externalities: Does the size matter?" 2014/38, Ferraresi, M.; Rizzo, L.; Zanardi, A.: "Policy outcomes of single and double-ballot elections"



Des campus d'envel·lèncie in BIKC 🚟 🔤 🛄



ieb@ub.edu www.ieb.ub.edu

Fiscal Federalism